

Temi di discussione

(Working Papers)

Bank resolution and public backstop in an asymmetric banking union

by Anatoli Segura Velez and Sergio Vicente







Temi di discussione

(Working Papers)

Bank resolution and public backstop in an asymmetric banking union

by Anatoli Segura Velez and Sergio Vicente

Number 1212 - March 2019

The papers published in the Temi di discussione series describe preliminary results and are made available to the public to encourage discussion and elicit comments.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

Editorial Board: Federico Cingano, Marianna Riggi, Emanuele Ciani, Nicola Curci, Davide Delle Monache, Francesco Franceschi, Andrea Linarello, Juho Taneli Makinen, Luca Metelli, Valentina Michelangeli, Mario Pietrunti, Lucia Paola Maria Rizzica, Massimiliano Stacchini. *Editorial Assistants:* Alessandra Giammarco, Roberto Marano.

ISSN 1594-7939 (print)

ISSN 2281-3950 (online)

Printed by the Printing and Publishing Division of the Bank of Italy

BANK RESOLUTION AND PUBLIC BACKSTOP IN AN ASYMMETRIC BANKING UNION

by Anatoli Segura[•] and Sergio Vicente^{*}

Abstract

This paper characterizes the optimal banking union with endogenous participation in a two-country economy in which domestic bank failures may be contemporaneous to sovereign crises, giving rise to risk-sharing motives to mutualize bail-out funding. Raising public funds to conduct bail-outs entails a deadweight loss. Bank bail-ins create disruption costs. The optimal resolution trades-off these costs. Truthfully eliciting information from domestic authorities imposes a domestic co-payment to fund bail-outs. When country asymmetry is large, ensuring the ex-ante participation of the fiscally stronger country requires a reduced contribution by this country, which increases the likelihood of bailing out its failing bank.

JEL Classification: G01, G21, G28.

Keywords: banking union, bail-in, bail-out, public backstop, mechanism design.

Contents

1. Introduction	
2. The baseline model	
3. Bank resolution in autarky	
4. Optimal banking union	
5. Discussion of the results	
6. Extensions	
7. Conclusion	
Appendix	
References	

^{*}Banca d'Italia, Directorate General for Economics, Statistics and Research.

^{*}Banca d'Italia and Queen Mary University in London.

1 Introduction¹

During the global financial crisis, a large number of banks around the world were intervened by resolution authorities. In the U.S., a few states concentrated a large share of the bank resolutions, with Georgia facing 87 bank shutdowns amounting to 24.6% of its banking system, Florida experiencing 70 bank closures corresponding to 21.8% of its banking system, and Puerto Rico undergoing the closedown of 3 banks comprising 30% of its banking system. The resolution of all these banks, which were mostly operating within the confines of their jurisdictions, was carried out at the federal level, with an estimated cost of \$72.5 billion to the Federal Deposit Insurance Corporation (FDIC).² Effectively, this injection of funds constituted a large source of redistribution of wealth from the most affluent territories to the states that experienced the largest shocks in their banking systems. In contrast, the public support programs for failing banks in the eurozone during the same period were funded at the national level. Several countries, chiefly Spain, Ireland, Greece, and Cyprus, had to inject massive amount of resources into their banking systems at a time at which their public finances were fragile, deepening their sovereign-debt crises. Although these countries obtained some financial assistance from other eurozone members to deal with the resolution of their failing banks, the degree of redistribution achieved through these programs was minimal, as most of the foreign support consisted of loans to the sovereigns with strong

¹The views expressed in this paper are our own and do not necessarily coincide with those of Banca d'Italia. We would like to thank Thorsten Beck, Arnaud Boot (discussant), Elena Carletti, Guillermo Caruana, Jean-Eduard Colliard, Giovanni Dell'Ariccia, Xavier Freixas (discussant), Pedro Gete, Florian Heider, Luc Laeven, Agnese Leonello (discussant), Gerard Llobet, Gyöngyi Lóránth, Robert Marquez, David Martinez-Miera, Martin Oehmke, Fausto Panunzi, David Pothier (discussant), Rafael Repullo, Andrzej Stopczyński (discussant), Javier Suarez (discussant), Xavier Vives, Ansgar Walther (discussant), Lucy White and seminar audiences at the 8th Oxford Financial Intermediation Conference (OxFIT), the CEPR 1st Endless Summer Conference on Banking and Corporate Finance, the CEPR 28 European Summer Symposium in Financial Markets (Gerzensee, evening sessions), the EBA 7th Annual Research Workshop–Reaping the benefits of an integrated EU banking market, the 10th Baffi Carefin Banking Regulation Conference, the 10th Lisbon Meetings in Game Theory and Applications, the II MadBar Workshop at Universitat Pompeu Fabra, the 1st ESCB Financial Stability Workshop, the 25th Meeting of the Spanish Finance Association, and the 42nd Simposio of the Spanish Economic Association. Sergio is grateful to Fundación UCEIF, Banco de España, and Ministerio de Economía, Industria y Competitividad (ECO2017-82385-P). Contact emails: Anatoli.SeguraVelez@bancaditalia.it; Sergio.Vicente@qmul.ac.uk.

²This amount does not include the cost incurred by the Treasury, the Federal Reserve, and the FDIC itself in assisting Wachovia, Citigroup, and Bank of America under the systemic risk exception decided on these three institutions. For further details, see "Crisis and Response: An FDIC History, 2008–2013." FDIC. 2017. Washington, DC.

conditionality to ensure its repayment.³

The European sovereign-debt crisis motivated the creation of a eurozone banking union with the three-fold objective of providing a common supervision, a common resolution, and a common fiscal backstop for the banks in the area.⁴ The first two objectives have been achieved with the establishment of the European Central Bank (ECB) and the Single Resolution Board (SRB) as central supervision and resolution authorities, respectively. The creation of the European Stability Mechanism (ESM) constitutes the first step towards the creation of a common fiscal backstop. However, the European Commission has issued a communication identifying "a last resort common fiscal backstop for the single resolution mechanism" as a critical missing element to complete the banking union.⁵ The International Monetary Fund (IMF), in turn, has also warned that the banking union currently falls short of providing sufficient risk-sharing across countries.⁶ However, several proposals to strengthen the common fiscal backstop have faced the opposition of the core countries of the eurozone under the concern that they may end up being net contributors to the common fiscal backstop.⁷

How should a common fiscal backstop be optimally designed? How much risk-sharing among countries can it provide? And how does the presence of a common fiscal backstop affect the resolution of banks? We address these questions in a model that takes into account two asymmetries that are prevalent in the context of the eurozone banking union. First, there exists a significant degree of heterogeneity on the fiscal strength of member states. Second, the newly created supranational resolution authority is required to operate in strict

³Financial support from the European Union was provided through the European Financial Stability Facility and the European Financial Stability Mechanism, two ad-hoc emergency programs that constituted the embryo of the current European Stability Mechanism.

⁴Euro Area Summit, 2012. Euro Area Summit Statement. Brussels, June 29.

 $^{{}^{5}}$ EC, 2017. Communication to the European Parliament, the Council, the European Central Bank, the European Economic and Social Committee and the Committee of the Regions on completing the banking union. European Commission 11/10/2017.

 $^{^{6}}$ IMF, 2016. 2016 article iv. consultation press release; Staff report; and statement by the executive director for the euro area. IMF Country Report No. 16/219.

⁷The following excerpt in the context of the establishment of a common fiscal capacity at the EU level provides an example: "The common fiscal capacity must not represent a disguised form of permanent transfers from one part of the union to another. Without effective safeguards in the design of the mechanism, which will ensure its fiscal neutrality amongst the Member States in the medium term, it could prove difficult to address the concerns of some Member States that they will become overall net contributors." (Interparliamentary conference on stability, economic coordination and governance in the European Union. Bratislava 16-18 October, 2016).

cooperation with more established domestic resolution authorities which, as put forward by Navaretti, Calzolari and Pozzolo (2015), may have informational advantages about the local banking sector.⁸ Our stylized model shows how fiscal and informational asymmetries shape the way a banking union should provide risk-sharing across countries. This analysis highlights the need for imposing laxer conditions on countries with initially stronger fiscal positions to access the common fiscal backstop, which increases the probability of bailing out their banks.

The environment consists of a two-date and two-country economy with one bank and a resolution authority in each country. The bank funds its assets issuing debt from local investors. If its assets do not deliver the promised return to debtholders, the bank is resolved through either a bail-out or a bail-in. Raising public funds to conduct a bail-out entails a deadweight loss, which may be thought of as the monetary cost of raising funds through distortionary taxation. A bail-in, which imposes losses on debtholders, generates a costly disruption in the economy where the bank resides. The resolution policy trades off these costs. When a country undergoing a sovereign crisis faces a bank failure, the recourse to the foreign country's funds may reduce the overall cost of funding a bail-out, because the cost of public funds is exceptionally high during a crisis. This risk-sharing motive constitutes the rationale for the creation of a banking union.

Precisely, a banking union consists of a set of policies to resolve a failing bank that are contingent on both the ex-ante standpoint of the sovereigns (i.e., the relative likelihood that countries will undergo a crisis at a future date) and the ex-post state of the sovereigns at the time of resolving the bank (i.e., whether countries are fiscally stable or undergoing a crisis). The paper characterizes the resolution policies that minimize the ex-post aggregate resolution cost subject to the ex-ante participation constraints of both countries. A standard mechanism-design approach to elicit the private information about the bail-in disruption costs allows us to focus on resolution mechanisms that establish the home and the foreign country's contribution to fund a bail-out and delegate the resolution decision to the home

⁸The Single Resolution Mechanism (SRM), whose purpose is to "ensure an orderly resolution of failing banks," is composed by the Single Resolution Board and the National Resolution Authorities of participating Member States. The Regulation No 806/2014 of the European Parliament, which establishes the procedures for the resolution of banks in the framework of the SRM, states that "National Resolution Authorities should assist the Board in resolution planning and the preparation of resolution decisions." See Section 2.1 for a further discussion of the informational asymmetries between domestic and supranational authorities.

authority.

The paper first characterizes the *ex-post optimal banking union*, namely the set of resolution policies that minimizes the aggregate resolution cost without considering the countries' participation constraints. In the event in which the home country is undergoing a crisis and the foreign country is fiscally stable, the ex-post optimal resolution policy establishes a co-funding of bail-outs. The home country copayment is a consequence of the informational asymmetry between the supranational and the national authorities. Requiring the home country to partially finance the bail-out increases the overall resolution cost. However, it reduces the home authority's willingness to overstate the magnitude of the bail-in contagion cost to persuade the central authority of the need for a bail-out, which would be partially funded by the foreign country.⁹ In the rest of contingencies (e.g., when both countries are simultaneously stable) there is no cost gain from co-funding bail-outs. Moreover, any foreign aid to fund bail-outs would increase the willingness of the sovereigns to bail out its failing banks. Consequently, the ex-post optimal resolution rules prescribe that the home sovereigns fully fund their own bail-outs in these contingencies.

The ex-post optimal banking union minimizes the aggregate resolution cost, but induces a positive expected net transfer from the stronger country to the weaker country, since the latter is more likely to experience a sovereign crisis. As a result, the ex-post optimal banking union is feasible only when countries' fiscal strength is not too dissimilar. When countries are sufficiently asymmetric, the *optimal (feasible) banking union* distorts the ex-post optimal resolution policy in two dimensions to ensure the voluntary participation of the fiscally stronger country.¹⁰ First, in the contingencies in which the ex-post optimal mechanism prescribes a copayment of bail-outs, the stronger country both reduces its contribution to fund a bail-out of a foreign bank and receives a larger aid to fund a bail-out of its home bank. Second, the stronger country receives partial funding for its bail-outs in the contingencies in which there is no cost advantage from sharing the bail-out cost (e.g., when both countries are concurrently fiscally stable). These country-specific contributions to the funding of bail-outs

⁹The structure of this copayment scheme resembles the out-of-pocket deductible displayed by several insurance contracts to reduce moral hazard. In this framework, however, the role of the copayment is to elicit the private information that domestic authorities possess.

¹⁰A particular case in which countries are very asymmetric is the situation in which the strong country is always (expected to be) fiscally stable. We address this case in Section 4.3 (ii) and show that even in this case an optimal common fiscal backstop strictly improves upon autarky as long as the fiscally weak country is not always (expected to be) undergoing a fiscal crisis.

effectively imply laxer conditions for the stronger country to access the common fiscal backstop. The optimal resolution framework thus increases the likelihood that banks in fiscally stronger countries get bailed-out, which suggests that the minimization of the aggregate resolution cost may not be compatible with leveling the playing field among banks within a banking union with heterogeneous countries.¹¹ Notice that although these conditions seem to privilege the stronger country, the weaker country appropriates all the expected welfare gains that the banking union creates. Therefore, forming a banking union does not affect the expected aggregate welfare of the fiscally stronger country. However, since its banks are more likely to be bailed out, the banking union effectively leads to a redistribution of welfare from the taxpayers to the banks' owners.

One of the obstacles towards the completion of the eurozone banking union has been the core countries' concern that it may lead to net transfers of funds towards the periphery. On the contrary, the optimal banking union features a net expected transfer from the fiscally weaker country towards the fiscally stronger when the country asymmetry is large. In this case, while the stronger country benefits from an expected positive net transfer inflow, the weak country obtains the strong country's aid when it is most valuable, that is, when it concurrently experiences a sovereign crisis and a bank failure. This result highlights that imposing fiscal neutrality for each member country constitutes a suboptimal way of ensuring each country's legitimate demand that the creation of a banking union is not detrimental to its constituency.

The baseline model omits several relevant aspects of a banking union that we address as extensions of the basic framework. In the first one, the banking union includes the prerogative of the supranational authority to early intervene banks upon the observation of a signal of the quality of their assets. An intervention ahead of the potential deterioration of a bank's assets allows repaying the bank's creditors without resorting to public funds. When countries asymmetry is sufficiently large, the optimal early intervention policy features forbearance with the stronger country's bank, allowing the continuation of the bank under some contingencies in which it would be efficient to shut it down. The rationale for this ex-post suboptimal early intervention policy is that it softens the fiscally stronger country's

¹¹Acharya (2003) also finds that creating a "leveled playing field" in capital requirements across countries may confer a competitive advantage to the banks in the countries with the most forbearing approach to bank resolution.

participation constraint by reducing its expected net transfer outflow in the resolution phase.

A second extension enlarges the set of bail-out funding tools to include the possibility of granting a bridge loan between sovereigns. Since loans are paid back at a future date, once the country undergoing a crisis may have regained its fiscal stability, they constitute a less costly form of copayment than contemporaneous home contributions. However, the optimal funding mix prescribes the use of direct transfers from a stable foreign country as well, as these funds constitute the least costly source of funding available within the banking union. This finding stands in contrast to the common fiscal backstop provided by the ESM, which considers direct transfers to be a last recourse option.¹²

A final extension introduces cross-border liabilities in our framework and shows that the cross-country ownership of banks' debt partially realigns the home and the foreign countries' objectives, enhancing the welfare gains induced by a banking union. The results are consistent with the view that an important driver for the creation of a banking union is the need to coordinate the supervision and the resolution actions for banks with significant cross-border liabilities.¹³

1.1 Related Literature

Our paper contributes to several strands of the literature. First, it belongs to an emerging literature on the architecture of banking unions that addresses the implications of the division of prerogatives between supranational and domestic authorities. In Foarta (2018), a supranational authority allocate resources to fund bail-outs, but domestic policymakers can divert political rents. As a result, there is an excessive expenditure in inefficient bail-outs, which may reduce welfare as compared with an autarky benchmark. Calzolari, Colliard, and Loranth (2017) highlight the unintended consequences of centralizing the supervision of multinational banks, which reduces the coordination problems among national agencies, but may inefficiently induce banks to adjust their international organizational structure to reduce supervisory monitoring. Carletti, Dell'Ariccia, and Marquez (2016) build a model of a "hub-and-spokes" banking union in which a central authority takes decisions based on

¹²Arts. 15-20 of the ESM Treaty.

¹³For instance, see van Claessens and van Horen (2013) for a survey on cross-border banking, Bolton and Oehmke (2018) on the resolution of cross-national banks, and Schoenmaker (2011) for a critical assessment of the impossibility of simultaneously achieving financial stability, financial integration, and national financial policies.

the information revealed by local supervisors. They find that a tough central authority may weaken the local authority's incentives to collect information, thereby increasing risk-taking by regulated banks. Similarly, in Colliard (2018) the central authority has an informational disadvantage relative to the local authority. The paper identifies a complementarity between the centralization of supervision and the degree of financial integration, which may lead to inefficient equilibria with local supervision and low financial integration. A common feature in these papers, which is also central in our analysis, is that the asymmetric information and the misalignment of preferences between domestic and supranational authorities leads to distortions of the resulting banking unions. Relative to these papers, our work is—to the best of our knowledge—the first one to analyze how the mutualization of a public backstop affects national authorities' incentives to reveal all their relevant information to the supranational resolution authority, and its implications for risk-sharing among the members of the banking union.

Several papers analyze how the heterogeneity across jurisdictions affects the feasibility and efficiency of supranational banking agreements. In Dell'Ariccia and Marquez (2006), competing domestic regulators do not internalize cross-border externalities, which leads to inefficiently low regulatory standards. Centralized regulation with common standards enhances overall welfare, but is only feasible when jurisdictions are sufficiently homogeneous. Similarly, in Beck and Wagner (2016) the establishment of centralized supervision with common standards reduces cross-border default externalities, but is not feasible when countries are very heterogeneous. Bolton and Oehmke (2017) analyze the choice of a single-point-ofentry (SPOE) versus a multiple-point-of-entry (MPOE) for the resolution of global banks. They find that although the SPOE resolution is more efficient, it is not implementable when the expected transfers across jurisdictions are too asymmetric. Our work departs from these papers in that we analyze the country-specific arrangements that are both feasible and maximize aggregate welfare when countries are largely asymmetric, finding that country heterogeneity must translate into an asymmetric treatment of countries. In this respect, we are closer to Stavrakeva (2018), who argues that capital requirements should be lower in countries with larger fiscal capacities because they are better equipped to conduct bail-outs in their banking systems.

Our work is also related to a broader literature that focuses on risk-sharing agreements

among heterogeneous countries. Tirole (2015) analyzes how the optimal mutual insurance pact among countries that may experience a sovereign crisis depends on their degree of asymmetry. Similarly, Fahri and Werning (2017) address the problem of country risk-sharing within an optimal currency area and characterize the optimal transfers as a function of the countries' characteristics. In a similar vein to these papers, we establish the voluntary risksharing agreements that can be reached through a banking union in which fiscally strong countries demand an asymmetric treatment to avoid subsidizing weaker countries.

Finally, this paper is also related to a recent theoretical literature that assesses the effects of different approaches to bank resolution. Dell'Ariccia et al. (2018) examine the costs and benefits of bail-ins and bail-outs, hinging on the trade-off between the moral hazard costs associated with bail-outs and the potential spillovers arising from bail-ins. Colliard and Gromb (2018) show that the use of bail-in tools in bank resolution may reduce the bank stakeholders' incentives to restructure bank debt. Walther and White (2017) find that a commitment to impose bail-ins based on private information often lacks credibility, because regulators must avoid conveying bad news that may spread over to the financial system. Our paper differs from these in that our focus is the impact of the supranational-versusdomestic informational asymmetry and the country heterogeneity in the implementation of the resolution policy.

The rest of the paper is organized as follows. Section 2 presents the model set-up, including an additional subsection to discuss some of the assumptions of the model. Section 3 describes the resolution decisions in the absence of a banking union. Section 4, which constitutes the core of the paper, characterizes the optimal design of the banking union and shows how it depends on the degree of country asymmetry. Section 5 provides a discussion of some policy relevant features of the main results. Section 6 conducts the extensions of the baseline model. Section 7 concludes. The proofs of the formal results of the paper are contained in the Appendix.

2 The baseline model

In this section we describe the model setup. In Section 2.1 we discuss some of the main assumptions of the model. We consider a two-country– $i \in \{1, 2\}$ –and two-date– $t \in \{0, 1\}$ – economy where all agents are risk-neutral and have a zero discount rate. We refer to country 1 as the *strong* country and we label country 2 the *weak* country. As we formalize below, this distinction stands for the higher likelihood that the weak country undergoes a sovereign crisis. In each country there is a bank, a continuum of competitive investors, and a local bank resolution authority. We refer to the bank and the local authority in country i as bank i and authority i, respectively. We also say that authority i is the home authority of bank i and we use the notation \overline{i} to refer to the other country, or to its agents.

Resolution authorities aim at maximizing the aggregate expected utility in their respective jurisdictions. They have resolution prerogatives over their country's bank and the capability to raise public funds to implement their initiatives. In addition, these domestic resolution authorities may agree at the initial date on the creation of a banking union to confer their prerogatives to a central resolution authority, as we specify below.

At t = 0, each bank has access to a project that requires an outlay of 1 unit. Banks raise these funds from local investors, issuing debt claims with face value D_i . The project's payoff at t = 1 is R > 0, if successful, and 0 otherwise. The probability that the project succeeds is p.

If a bank's project fails at t = 1, the bank cannot repay its debt and the competent resolution authority must resolve the bank. There are two types of resolution: A *bail-out* and a *bail-in*.

Bank bail-out A bank bail-out consists of the repayment of the unit of principal investment to the bank's debtholders.¹⁴ Bail-outs are funded with public resources. We assume that the net social cost of public funds at t = 1 is λ_S if the country is fiscally *stable* (S) and λ_C if the country is undergoing a sovereign *crisis* (C), with $0 < \lambda_S < \lambda_C$. For each pair $\sigma \in \{SS, SC, CS, CC\}$ representing the state of the public finances in each of the countries at t = 1, we denote by q^{σ} the probability as of t = 0 that state σ realizes. We assume that the state of both countries public finances is common knowledge at t = 1. We denote q_i^S and $q_i^C \equiv 1 - q_i^S$ the probability that country *i* is stable and undergoing a crisis at t = 1, respectively. Hence, we have that $q_1^S = q^{SS} + q^{SC}$ and $q_2^S = q^{SS} + q^{CS}$. We assume

¹⁴We assume that debtholders only get the principal 1 for simplicity, as it renders the resolution decisions and the expected social costs of resolution independent of the promised repayment D_i . Alternatively, we could consider that a bail-out consists of the repayment of the debt face value D_i . This assumption would introduce a dependence of the cost of bail-outs on bail-out expectations, which would complicate the analysis without adding further insight.

that the strong country (i = 1) is more likely to be stable than the weak country (i = 2), that is:

Assumption 1 (Strong country more likely to be fiscally stable) $q_1^S > q_2^S$.

Notice that Assumption 1 implies both $q^{SS} + q^{CC} < 1$ and $q^{SC} > 0$, which implies that countries' public finances are not perfectly correlated.¹⁵

Bank bail-in A bank bail-in imposes the full burden of bank losses on debtholders. We assume that the loss suffered by bank debtholders may disrupt the rest of the financial sector, and therefore the economy at large, and generate a local deadweight contagion cost of k. As of t = 0, k is a random variable, which we assume to be uniformly distributed in the interval $[0, k_{\text{max}}]$. For notational simplicity, we write its probability density function and cumulative distribution function as g(k) and G(k), respectively. The bail-in cost k is realized at t = 1, after the bank's project fails, but before the resolution decision is taken. We assume that its realization is only observed by the home authority. We make the following assumption:

Assumption 2 (Inefficient unconditional bail-outs) $k_{\max} > \lambda_C > \lambda_S > \int_0^{k_{\max}} k dG(k)$.

The assumption that $\lambda_S > \int_0^{k_{\text{max}}} k dG(k)$ implies that unconditionally bailing out a bank is inefficient, even if the funding of the bail-out is entirely provided by a stable country. Put differently, this assumption asserts that in the absence of information on the realized value of k the optimal choice is to bail in a failing bank, regardless of the cost of public funds. This assumption is consistent with the Bank Recovery and Resolution Directive (BRRD) adopted by the EU, which attempts to make bail-ins the norm in bank resolutions, bail-outs being the exception.¹⁶ The assumption that $k_{\text{max}} > \lambda_C$ is inessential and only imposed to

¹⁵Observe that Assumption 1 does *not* rule out the particular case that the strong country is always stable, that is, $q_1^S = 1$. Indeed, we analyze this special case in Section 4.3.

¹⁶The Bank Recovery and Resolution Directive (BRRD)–Directive 2014/59/EU of the European Parliament and of the Council of 15 may 2014 establishing a framework for the recovery and resolution of credit institutions and investment firms–provides a common framework for the resolution of failing banks and confers authorities more powers to impose losses on private stakeholders, the so-called bail-in tools. Although the norm in the BRRD is to resolve banks through bail-ins, this directive allows public support to failing institutions when authorities consider that private burden sharing would endanger financial stability. However, the BRRD establishes strong conditions before resorting to public funds: The bank's shareholders and creditors must bear losses of at least 8% of all liabilities, and the privately funded Single Resolution Fund must contribute 5% of all liabilities to the resolution of the bank.

reduce the number of cases to consider.¹⁷ It implies that bail-outs are preferable to the most disruptive bail-ins, even if the net social cost of public funds is high.

Banking Union A bank failure may occur when its home country is experiencing a crisis and the foreign country is stable. Consequently, domestic authorities may find it convenient to provide each other with some insurance by agreeing at the initial date to mutualize (to some extent) the funding of public support to failing banks. More precisely, we assume that at t = 0 local authorities may create a banking union conferring their prerogatives (both the decision on the resolution of failing banks and the capability to levy resources from sovereigns) to a central authority. This central authority operates without information on the realized bail-in contagion costs so that it must resort to the local authorities to elicit such information. The banking union agreement establishes a protocol of bank resolution decisions and sovereign contributions to bail-outs in each contingency that may arise.

Formally, a banking union is characterized by a resolution mechanism $\{b_{i,\sigma}(\hat{k}), x_{i,\sigma}(\hat{k})\}$, where the bank failure state (i, σ) represents the location $i \in \{1, 2\}$ of the failing bank and the state $\sigma \in \{SS, SC, CS, CC\}$ of the public finances in both countries at t = 1. For each disruption cost \hat{k} reported by the failing bank's home authority, the resolution mechanism establishes whether the bank is bailed out $(b_{i,\sigma}(\hat{k}) = 1)$ or bailed in $(b_{i,\sigma}(\hat{k}) = 0)$, as well as the home country contribution $x_{i,\sigma}(\hat{k}) \in [0, 1]$ in case a bail-out is prescribed, the remaining funding needs $1 - x_{i,\sigma}(\hat{k})$ being contributed by the foreign country.

By the virtue of the revelation principle, we can focus on truth-telling resolution mechanisms for each bank failure state (i, σ) . Using standard mechanism design arguments, we show in Appendix A that any relevant truth-telling mechanism that elicits the cost k and establishes a contingent bail-out/bail-in decision can be implemented as follows: The central authority sets the home contribution $x_{i,\sigma} \in [0, 1]$ to the funding of a bail-out and delegates the decision of whether to bail out or bail in the bank to the home authority. Consequently, we can characterize a banking union by a vector $\boldsymbol{x} = (x_{i,\sigma})_{i \in \{1,2\}, \sigma \in \{SS, SC, CS, CC\}}$, establishing the home contribution to bail-outs in each of the eight bank failure states (i, σ) . Note that the degenerate "banking union" described by the vector of home contributions $\mathbf{1} = (1, ..., 1)$ corresponds to the autarky case, in which each country resolves its bank failures on its own.

¹⁷It suffices to have $k_{\text{max}} > \lambda_S$ so that bail-ins are not always preferable to bail-outs.

From now on, we will always refer to a banking union x as to a proper banking union, that is, one in which $x \neq 1$.

Finally, we say that a banking union x is *feasible* if each country attains at least as much expected utility as in autarky. The banking union problem we address in the paper is that of analyzing the optimal banking union, that is, the aggregate welfare maximizing banking union that ensures the endogenous participation of the two countries.

2.1 Discussion of assumptions

In this section we discuss some of the assumptions made in the model.

Reduced form for bail-in disruption costs

The establishment of a new bank resolution framework within the eurozone, which includes the so-called bail-in tools, is paving the way towards minimizing the negative spillovers of bank bail-ins. However, several authors have warned that in some situations the use of bail-in tools may still cause severe disruptions in the rest of the economy (e.g., Dewatripont, 2014; Goodhart and Avgouleas, 2016). Systemic spillovers may manifest in different forms. For instance, bank bail-ins may undermine the investors' confidence, leading to roll-over freezes or deposit withdrawals, which may compromise financial stability (see Brown, Evangelou, and Stix, 2018, for evidence on this phenomenon following a bail-in event in Cyprus in 2013). Besides, as put forward by Dell'Ariccia et al. (2018), bail-in spillovers may be mechanically transmitted through the balance sheet of the bank, endangering the solvency of its debtholders and precipitating chains of bankruptcies. As evidenced by Beck, Da Rocha Lopes, and Silva (2017), which analyze the response of financial institutions to a bail-in in Portugal in 2014, the banks holding the bailed-in debt may tighten the access to credit to its borrowers, leading to further deterioration of the economic outlook. We model all such potential spillovers in reduced form, subsuming them in the parameter k. Finally, we assume for the sake of simplicity that the distribution of the bail-in disruption cost is independent of the state of the sovereigns. We could instead adopt the arguably more realistic assumption that the bail-in cost distribution is shifted to the right when the failing bank's home country is undergoing a crisis with no substantial changes in the qualitative results.

Informational advantage of national authorities

We assume that national authorities possess better information than the supranational authority about the potential consequences that a bank bail-in may inflict on the country's economy. This information advantage may arise from several sources. First, the SRB is a newly created authority, while national resolution authorities have been operating for a longer time and may have accumulated better knowledge about the domestic banking sector. Such expertise may allow them to better assess the relevant determinants of the bail-in spillovers. Second, and relatedly, the SRB is a resolution authority that is formally independent of the ECB, which is the supranational supervisory authority in the banking union. The SRB can only gather information on banks through the ECB-or the relevant national supervisory authorities-through a memorandum of understanding between the two institutions. Instead, in several countries the resolution and the supervision authorities, although organically different, reside in the central bank, which has access to bank data to conduct its supervisory function. This privileged access to information on the entire domestic banking system might prove especially useful in the short time windows under which bank resolution processes are typically conducted. Third, in the similar context of the bank-supervisor relationship, several empirical analyses show that information acquisition is better the closer the geographical and cultural proximity between banks and their supervisors. Gopalan, Kalda, and Manela (2017) find that a larger geographical distance reduces the supervisor's effectiveness in reducing the bank's risk-taking behavior. Along these lines, Bian, Haselmann, Kick, and Vig (2015) state that "[...] proximity [between supervisors and banks] has the potential to improve the decision-making process, as it provides politicians with good information about banks that get into distress." Fourth, although the transposition into national law of the BRRD has constituted an important step towards the harmonization of the legal framework governing bank resolutions in the EU, "the divergence of national insolvency laws is a major obstacle towards a fully-fledged banking union", as recently acknowledged by the Chair of the SRB.¹⁸ The particularities of national insolvency laws do in fact play a key role in determining the counterfactual of "no-creditor-worse-off" required for an application of the bail-in tools that is not exposed to potentially very costly litigation actions from affected bank stakeholders. In this respect, national authorities are likely to have a deeper knowledge of the intricacies of the domestic insolvency law due to the expertise of their human capital and the experience built throughout time operating under this legal framework. Finally, as argued by Ferrarini (2015), the institutional set-up of the eurozone banking union exhibits "semi-strong" centralization, consisting on supranational authorities with prerogatives on banks supervision and resolution, but sharing their tasks and operation in coordination with national authorities. This architecture may well constitute the optimal institutional response to the informational advantages of national authorities, at least in the current juncture.

Transfers across countries only for funding bail-outs

The paper restricts transfers across countries within the banking union to be used to fund

¹⁸Eurofit article by Elke König: "Real defragmentation of the Banking Union: the way forward," Thursday, 26 April 2018, https://srb.europa.eu/en/node/544

bail-outs of failing banks. We could allow unconditional transfers among countries, that is without any restriction on the use of these funds. With these type of transfers, fiscally strong countries could be compensated by weak countries at the inception of the banking union with independence of their banking sector needs. However, this sort of transfers have not been considered in the policy debate. Indeed, several voices have explicitly opposed integrating the banking union common fiscal backstop in the common EU budget, which would effectively delink the funding of bank bail-outs from the countries' contribution to the EU budget.¹⁹ To better understand the mechanisms driving the main results of the paper, in Section 5.2 we address the implications of relaxing the assumption that transfers be only used to fund bail-outs.

National authorities' preferences

As noted by Navaretti, Calzolari and Pozzolo (2015), the assistance of national resolution authorities to the SRB on the preparation of resolution plans potentially creates a conflict of interest or coordination problems between national and central authorities, since their objectives are likely not entirely aligned. Along these lines, we assume that each domestic supervisor is concerned only with the welfare of the agents within its constituency, as in Colliard (2018), Carletti, Dell'Ariccia and Marquez (2016), and Calzolari, Colliard and Loranth (2018).²⁰ This view is consistent with the long-standing political economy view that national agents neglect the impact of their actions on foreign countries, recently put forward by Faia and Weder di Mauro (2015). Nonetheless, all of our qualitative results hold if we assume that national authorities also care for the well-being of the agents of the banking union at large, as long as they attribute a higher weight to their respective constituencies.

Cross-border liabilities

The main rationale for the creation of a banking union within the framework of this paper is risk-sharing across countries with independent banking systems. Many commentators have

¹⁹In fact, as evidenced by several agreements among eurozone countries, there exists a strong political opposition to build on financial capabilities to address uncertain episodes of financial distress. Common public funds to support a sovereign in distress are typically only mobilized once the dreadful event has already materialized and only employed to resolve the particular problem at hand. For further details, see "Strengthen stability: further development of the ESM in a deepened monetary union," by Klaus Regling, ESM Managing Director, at the Parliamentary evening of the association of German savings banks on 17 October 2018.

 $^{^{20}}$ The misalignment of objectives within the US banking system has been put forward by Agarwal *et al.* (2014), who find that federal regulatory supervisors are systematically tougher than their state counterparts.

argued that the banking union should also serve the purpose of providing a common space for an increasingly integrated banking system. We take on this issue in Section 6.3 and show that the main insights from the baseline analysis are reinforced in a set-up in which there are cross-border debt-holdings.

3 Bank resolution in autarky

In this section we characterize the optimal resolution policy of a country that addresses its bank failures on its own. The associated expected utility of its agents constitutes the outside option for the country when deciding whether or not to participate in a banking union.

Consider first the situation in which the failure of a bank at t = 1 occurs when the home country is stable. A bail-out amounts to a redistribution of one unit of funds within local agents at a (net) social cost of $\lambda_S \cdot 1$. Hence, the home authority bails out the bank when the bail-in contagion cost exceeds the social cost of the bail-out, that is, when $k \ge \lambda_S$. The expected social cost of a bank resolution when its home country is stable is thus given by:

$$\Pi^{S} = \underbrace{\int_{0}^{\lambda_{S}} k dG(k)}_{Bail-in\ cost} + \underbrace{\lambda_{S}\left(1 - G\left(\lambda_{S}\right)\right)}_{Bail-out\ cost}.$$
(1)

If the bank's home country is undergoing a crisis the social cost of funds is $\lambda_C > \lambda_S$, which increases the cost of bail-outs and, consequently, elevates the bail-out region to $k \ge \lambda_C$. The expected social cost Π^C of a bank resolution when its home country is undergoing a crisis is then given by an expression analogous to (1) in which λ_S is replaced by λ_C . The expected cost difference of a bank resolution in these two states is therefore given by:

$$\Pi^{C} - \Pi^{S} = \underbrace{\int_{\lambda_{S}}^{\lambda_{C}} (k - \lambda_{S}) \, dG(k)}_{Bail-in \ versus \ bail-out} + \underbrace{(\lambda_{C} - \lambda_{S}) \left[1 - G(\lambda_{C})\right]}_{Cost-difference \ in \ bail-outs} > 0.$$
(2)

Resolution costs in a country undergoing a sovereign crisis increase relative to a stable country because the higher cost of public funds leads to some bail-ins that are especially costly (i.e., for $\lambda_S \leq k < \lambda_C$, as captured by the first term of the latter expression) and makes more costly the funding of bail-outs whenever they are conducted (i.e., for $k \geq \lambda_C$, as captured by its second term). Taking into account that banks' debt is priced competitively,²¹ the expected utility as of t = 0 of the agents in country *i*, which we denote \overline{U}_i , is given by:

$$\overline{U}_{i} = \underbrace{pR-1}_{Bank\ NPV} - \underbrace{(1-p)\left(q_{i}^{S}\Pi^{S} + q_{i}^{C}\Pi^{C}\right)}_{Resolution\ cost}.$$
(3)

The first term in this expression accounts for the expected value of the bank's project, while the second captures the expected resolution cost.

4 Optimal banking union

In this section we characterize the optimal banking union, that is, the aggregate welfare maximizing banking union within the class of agreements that ensure the participation of the two countries at t = 0. We first characterize the ex-post optimal resolution policies in a banking union. Then, we show that these are feasible if and only if the degree of asymmetry between the two countries is not too large. Finally, we analyze how resolution policies in the optimal banking union with large country asymmetry differ from ex-post optimal policies.

4.1 Ex-post optimal resolution policies

In this section, we describe the ex-post optimal resolution policies in a banking union for each state in which a bank may fail. The optimal resolution policies are defined as the aggregate welfare maximizing resolution policies that can be designed at t = 1 by a central authority with the prerogative of raising funds from the two countries, but without information on the realization of the bail-in contagion costs.

We start with the analysis of the case in which a bank fails at a country that is undergoing a sovereign crisis, while the other country is stable, which corresponds to the bank failure states (1, CS) and (2, SC). A resolution policy is characterized by the home contribution xto the funding of a bail-out. Observe that, by construction, the ex-post optimal resolution policy is independent of whether the home country is strong or weak.

 $pD_{i}^{*} + (1-p)\left[q_{i}^{S}\left(1 - G\left(\lambda_{S}\right)\right) + q_{i}^{C}\left(1 - G\left(\lambda_{C}\right)\right)\right] \cdot 1 = 1.$

²¹Notice that the equilibrium promised repayment D_i^* required by the competitive debtholders anticipates the probability of a bail-out, so that it satisfies:

The particular value of such promised repayment does not have welfare implications. We henceforth omit it from our computations.

We first address the effect of the required bail-out funding home contribution x on the home authority decision to ask for either a bail-out or for a bail-in after observing that the bail-in disruption cost amounts to k. Bailing out the bank entails a utility for the agents in the home country of $1 - (1 + \lambda_C)x$. The first term corresponds to the transfer to the failing bank's debtholders associated with the bail-out. The second term stands for the gross social cost of the resources that the home country is required to contribute to the bail-out. Bailing in the bank, in turn, entails a utility of -k. Comparing the two, we have that the home authority prefers to ask for a bail-out whenever the bail-in disruption cost k is above a threshold $\overline{k}(x)$ given by:

$$\overline{k}(x) = \max\{(1+\lambda_C)x - 1, 0\}.$$
(4)

Note that the bail-out threshold $\overline{k}(x)$ is increasing in x because higher home contributions make bail-outs more costly for the home country. In addition, $\overline{k}(x) = 0$ for all $x \leq \frac{1}{1+\lambda_C}$ and $\overline{k}(1) = \lambda_C$, since the home contribution to bail-outs x = 1 corresponds to the autarky situation.

The expected aggregate resolution cost associated with a home contribution x to bail-outs is therefore given by:

$$\Pi(x) = \underbrace{\int_{0}^{\overline{k}(x)} k dG(k)}_{Bail-in\ cost} + \underbrace{(\lambda_{S} + (\lambda_{C} - \lambda_{S})x)(1 - G(\overline{k}(x)))}_{Bail-out\ cost}.$$
(5)

Notice that the term including the expected aggregate cost of the funding of bail-outs takes into account the social cost of the funds contributed by each of the countries.

The ex-post optimal resolution policy corresponds to the home contribution $x^* \in [0, 1]$ that minimizes $\Pi(x)$. Notice that Assumption 2 implies that any home contribution leading to an unconditional bail-out (i.e., $x \leq \frac{1}{1+\lambda_C}$) is dominated by the autarky solution (i.e., x = 1).²² We can hence focus on the interval $x \in \left(\frac{1}{1+\lambda_C}, 1\right]$. In order to understand the drivers of the central authority decision, consider the marginal effect of an increase in the home contribution x on the expected aggregate cost of resolution $\Pi(x)$. Differentiating (5) yields:

$$\Pi'(x) = \underbrace{\left(\lambda_C - \lambda_S\right) \left[1 - G\left(\overline{k}(x)\right)\right]}_{Cost \ effect \ (+)} - \underbrace{\left(1 + \lambda_S\right) \left(1 - x\right) \left(1 + \lambda_C\right) g(\overline{k}(x)\right)}_{Overstatement \ effect \ (-)}.$$
(6)

 22 The formal argument is contained in the proof of Proposition 1.

The first term in (6) is a *cost effect*, which accounts for the increase in the social cost of funding the bail-out by requiring the home country undergoing a crisis to contribute a higher share. This effect is positive due to the wedge in the cost of public funds between the home and the foreign country, thus pushing down the optimal home contribution to bail-outs. The second term is an *overstatement effect*, accounting for the diminished willingness of the home country to call for a bail-out when its contribution to bail-outs increases. This effect is negative, pushing up the optimal home contribution to the funding of bail-outs.

The overstatement effect appears because the home authority does not fully internalize the cost of a bail-out. In particular, the aggregate cost of funding a bail-out is given by $\lambda_S + (\lambda_C - \lambda_S)x$, while the home authority only faces a cost $\overline{k}(x)$ -c.f. Expression (4) above. The wedge between the two, which is given by $(1 + \lambda_S)(1 - x)$, vanishes when the home country is required to fund bail-outs in its entirety (i.e., when x = 1). As a consequence, we have that $\Pi'(1) > 0$, which implies that $x^* < 1$. Hence, despite the informational frictions, the ex-post optimal resolution policy achieves some risk-sharing across countries that reduces the aggregate resolution cost when the home country is undergoing a crisis and the foreign country is fiscally stable.

For each of the other bank failure states the cost of public funds in the home country is not higher than in the foreign country so that there are no risk-sharing motives to mutualize the funding of bail-outs. An analogous analysis shows that in those states it is ex-post optimal that the home country of a failing bank funds its bail-out in its entirety.

The following proposition summarizes our discussion in this section. The formal proof is stated in the Appendix.

Proposition 1 (Ex-post optimal banking union) The vector $\mathbf{x}^* = (x_{i,\sigma}^*)$ of home contributions to bail-outs characterizing the ex-post optimal banking union satisfies:

(i) When the failing bank is located in a country undergoing a crisis, and the foreign country is stable, that is, in the bank failure states $(i, \sigma) \in \{(1, CS), (2, SC)\}$, the home contribution is given by $x_{i,\sigma}^* = x^* \in \left(\frac{1}{1+\lambda_C}, 1\right)$, which is uniquely characterized by the following condition:

$$\Pi'(x^*) = 0. \tag{7}$$

(ii) In any other bank failure state, that is, for $(i, \sigma) \notin \{(1, CS), (2, SC)\}$, home countries fund bail-outs in their entirety, that is, $x_{i,\sigma}^* = 1$.

4.2 Feasibility of ex-post optimal resolution policies

In this section we ask whether the ex-post optimal banking union x^* characterized in Proposition 1 satisfies the participation constraint of the two countries at the initial date, in which case x^* is the optimal banking union.

We denote $U_i(\boldsymbol{x}^*)$ the expected utility, as of t = 0, of the agents in country *i* within the banking union \boldsymbol{x}^* . Considering that the ex-post optimal resolution policies differ from those in autarky only in the bank failure states (1, CS) and (2, SC), in which the home contribution to bail-outs is $\boldsymbol{x}^* < 1$, it follows that the strong country's (positive or negative) net gain from joining the banking union \boldsymbol{x}^* is given by:

$$U_{1}(\boldsymbol{x}^{*}) - \overline{U}_{1} = \underbrace{(1-p)q^{CS} \left(\Pi^{C} - \Pi(\boldsymbol{x}^{*})\right)}_{Reduction \ of \ aggregate \ resolution \ cost} - \underbrace{(1-p) \left(q^{SC} - q^{CS}\right) \left(1 - \boldsymbol{x}^{*}\right) (1 + \lambda_{S}) \left(1 - G\left(\overline{k}(\boldsymbol{x}^{*})\right)\right)}_{Cost \ of \ net \ transfers \ from \ strong \ to \ weak \ country}$$
(8)

where \overline{U}_1 is as described in (3). The first term captures the gains attained by the creation of a banking union, which reduces the aggregate resolution cost when the strong country is undergoing a crisis and the weak country is stable (note that $\Pi(x^*) < \Pi(1) = \Pi^C$). The second term stands for the expected net transfers from the strong to the weak country implied by the foreign contribution to bail-outs $1 - x^*$ in states (1, CS) and (2, SC). Note that the term is proportional to the difference $q^{SC} - q^{CS}$ of the probabilities of these two states. Since $q^{SC} - q^{CS} = q_1^S - q_2^S > 0$, the expected net transfer is positive for country 1. Whenever the second term in (8) dominates the first one, the ex-post optimal banking union is not feasible, as the strong country prefers to remain in autarky.²³ The next proposition states that the ex-post optimal banking union is feasible if only if the degree of country asymmetry is not too large:

 $^{2^{3}}$ The expression for the weak country's gain from joining the ex-post optimal banking union can be obtained by swapping the direction of the difference $q^{SC} - q^{CS}$ in (8), which shows that the weak country is a net receiver of funds and thus always finds it beneficial.

Proposition 2 (Feasibility of the ex-post optimal banking union) There exists a degree of country asymmetry $a \in (0, 1)$ such that the optimal banking union exhibits ex-post optimal resolution policies if and only if $q^{CS} \ge aq^{SC}$.²⁴

4.3 Optimal banking union with large country asymmetry

In this section we characterize the optimal banking union when country asymmetry is large enough so that the ex-post optimal banking union does not meet the participation constraint of the strong country. In order to formalize the problem, we generalize the notation and objects introduced in Section 4.1. For any given pair of public finances σ at t = 1, we denote by $\lambda_{1,\sigma}$ and $\lambda_{2,\sigma}$ the social cost of funds in countries 1 and 2, respectively.²⁵ In addition, for a resolution policy described by a home contribution to bail-outs x, we denote by $\overline{k}(x|i,\sigma)$ the bail-out threshold that it induces, and we let $\Pi(x|i,\sigma)$ denote its expected aggregate resolution cost. These functions, which are defined analogously to (4) and (5), respectively, are given by:

$$\overline{k}(x|i,\sigma) = \max\left\{ (1+\lambda_{i,\sigma})x - 1, 0 \right\}, \tag{9}$$

$$\Pi(x|i,\sigma) = \int_0^{\kappa(x|i,\sigma)} k dG(k) + (\lambda_{\bar{i},\sigma} + (\lambda_{i,\sigma} - \lambda_{\bar{i},\sigma})x)(1 - G(\bar{k}(x|i,\sigma))).$$
(10)

It is also convenient to introduce the function $t(x|i, \sigma)$, which accounts for the expected gross social cost incurred by the foreign country (country \bar{i}) in contributing to the funding of bail-outs in the bank failure state (i, σ) . This function is given by:

$$t(x|i,\sigma) = (1-x)(1+\lambda_{\overline{i},\sigma})\left(1-G\left(\overline{k}(x|i,\sigma)\right)\right).$$
(11)

Let $\boldsymbol{x} = (x_{i,\sigma})$ be a banking union. We can write the expected utility at t = 0 of agents in country *i* within the banking union \boldsymbol{x} as follows:

$$U_{i}(\boldsymbol{x}) = \underbrace{pR-1}_{NPV} - \underbrace{(1-p)\sum_{\sigma} q^{\sigma} \left(\underbrace{\Pi(x_{i,\sigma}|i,\sigma)}_{Aggregate \ resolution \ cost} - \underbrace{t(x_{i,\sigma}|i,\sigma)}_{Cost \ of \ transfers \ from \ foreign \ country} \right)}_{Home \ resolution \ cost} - \underbrace{(1-p)\sum_{\sigma} q^{\sigma} t(x_{\overline{i},\sigma}|\overline{i},\sigma)}_{Cost \ of \ transfers \ to \ foreign \ country}}$$
(12)

²⁴We show in the proof of the proposition that $a \equiv \frac{(1-x^*)(1+\lambda_S)(1-G(\overline{k}(x^*)))}{\Pi^C - \Pi(x^*) + (1-x^*)(1+\lambda_S)(1-G(\overline{k}(x^*)))}$. ²⁵That is, $\lambda_{1,SC} = \lambda_{1,SS} = \lambda_{2,SS} = \lambda_{2,CS} = \lambda_S$ and $\lambda_{1,CS} = \lambda_{1,CC} = \lambda_{2,CC} = \lambda_{2,SC} = \lambda_C$. The second term in (12) stands for the expected (net) cost incurred by country i in the resolution of its own bank. This expression is obtained by subtracting the expected cost of transfers incurred by the foreign country from the aggregate resolution cost. The third term in (12) takes into account the expected cost for country i of transfers to fund bail-outs at the foreign country.

Adding up the corresponding expressions of (12) for countries 1 and 2 we can write the aggregate welfare in the banking union as follows:

$$W(\boldsymbol{x}) \equiv U_1(\boldsymbol{x}) + U_2(\boldsymbol{x}) = 2(pR - 1) - (1 - p) \sum_{(i,\sigma)} q^{\sigma} \Pi(x_{i,\sigma}|i,\sigma).$$
(13)

This function captures the NPV of the two bank projects, as well as the expected aggregate resolution costs in each bank failure state. Note that $t(x_{i,\sigma}|i,\sigma)$ and $t(x_{\bar{i},\sigma}|\bar{i},\sigma)$ cancel out, as they constitute transfers among countries.

The optimal banking union is the solution to the following maximization problem:

$$\max_{\boldsymbol{x}} \quad W(\boldsymbol{x}) \\ \text{s.t.} \quad U_1(\boldsymbol{x}) \ge \overline{U}_1 \ (PC_1) \ . \\ U_2(\boldsymbol{x}) \ge \overline{U}_2 \ (PC_2)$$
 (14)

It follows from (13) that solving the maximization problem in (14) is equivalent to minimizing the aggregate resolution costs, subject to the participation constraints of the two countries.

The following proposition provides necessary and sufficient conditions for a feasible banking union to exist when country asymmetry is large and provides the characterization of the optimal banking union in that case.

Proposition 3 (Optimal banking union with large country asymmetry) Let the degree of country asymmetry a be as defined in Proposition 2 and assume that $q^{CS} < aq^{SC}$ so that the ex-post optimal banking union is not feasible. Then, there exists a banking union that strictly increases aggregate welfare relative to autarky if and only if $q^{SC} < 1$. In such case, the optimal banking union $\hat{x} \equiv (\hat{x}_{i,\sigma})$ is unique and satisfies:

- (i) Only the participation constraint of the strong country binds, that is, $U_1(\hat{x}) = \overline{U}_1$ and $U_2(\hat{x}) > \overline{U}_2$.
- (ii) If a bank fails in a country that is undergoing a crisis, and the other country is stable, then the home contribution to a bail-out is smaller (resp. larger) than ex-post optimal if

the failing bank is located in the strong (resp. weak) country, that is, $\hat{x}_{1,CS} < x^* < \hat{x}_{2,SC}$. In addition, there is always foreign aid, even if the failing bank is located in the weak country, that is, $\hat{x}_{2,SC} < 1$.

- (iii) If a bank fails, and the state of the public finances of the two countries is the same, then there is foreign aid to fund a bail-out if and only if the failing bank is located in the strong country, that is, $\hat{x}_{1,SS} < \hat{x}_{2,SS} = 1$ and $\hat{x}_{1,CC} < \hat{x}_{2,CC} = 1$.
- (iv) If a bank fails in a country that is stable, and the other country is undergoing a crisis, then there is no foreign aid to fund bail-outs, that is, $\hat{x}_{1,SC} = \hat{x}_{2,CS} = 1$.

In order to shed light on the features of the optimal banking union characterized above, we discuss below a series of stylized economies in which, besides $\sigma = SC$, only one additional pair of public finances occurs with positive probability.

Optimal banking union in stylized economies with $q^{SC} \in (0, 1)$

(i) $q^{CS} > 0$ and $q^{SC} + q^{CS} = 1$ (Countries' public finances perfectly negatively correlated).

When the countries public finances are perfectly negatively correlated the strong country can obtain foreign aid in the bank failure state (1, CS) and contribute to the funding of bail-outs at the weak country in the bank failure state (2, SC). The proposition states that feasible risk-sharing across countries is optimally achieved by reducing the (more frequent) strong country foreign aid below the ex-post optimal level, and raising the (less frequent) weak country aid above the ex-post optimal level, that is:

$$0 < 1 - \hat{x}_{2,SC} < 1 - x^* < 1 - \hat{x}_{1,CS}.$$
(15)

The intuition is as follows. On the one hand, both distortions contribute to strictly softening the strong country's participation constraint. On the other hand, by construction the ex-post optimal policy minimizes the aggregate resolution cost—that is $\Pi'(x^*) = 0$ —so that distortions around that level of mutualization of the funding of bail-outs have a negligible impact on the aggregate resolution cost. As a result, it is optimal to distort both resolution policies.

(ii)
$$q^{SS} > 0$$
 and $q^{SC} + q^{SS} = 1$ (Strong country always stable).

If the strong country is always stable and the two countries are stable contemporaneously with positive probability, the strong country can obtain foreign support to bail out its bank in the latter contingency. In order to see that a banking union is feasible in this economy, consider the derivative of the aggregate resolution costs function in the bank failure state (1, SS), which is given by:

$$\Pi'(x_{1,SS}|1,SS) = \underbrace{(\lambda_S - \lambda_S) \left[1 - G\left(\overline{k}(x_{1,SS}|1,SS)\right)\right]}_{Cost \ effect \ (=0)} - \underbrace{(1 + \lambda_S) \left(1 - x_{1,SS}\right) \left(1 + \lambda_S\right) g(\overline{k}(x|1,SS)\right)}_{Overstatement \ effect \ (-)}.$$
(16)

Observe that foreign aid in state (1, SS) leads to an overall reduction of welfare, because the overstatement effect is at work and there is no cost gain. However, its impact on the overall welfare is small when the strong country home contribution $x_{1,SS}$ to a bail-out is large, that is, $\Pi'(x_{1,SS} = 1|1, SS) = 0$. The reason is that in this case the strong country internalizes to a large extent the cost of bailing out a home bank. Analogously, we can show that $U'_1(x_{1,SS} = 1|1, SS) > 0$, and thus a home contribution $\hat{x}_{1,SS} < x^*_{1,SS} = 1$ softens the strong country participation constraint with a negligible overall welfare effect. The strong country can thus be compensated for its contribution to the weak country bail-outs in state (2, SC) through foreign aid in state (1, SS). In addition, the optimal feasible risk-sharing requires that the support given by the strong country be below the ex-post optimal level, that is, $1-\hat{x}_{2,SC} < 1-x^*$. As explained above, the reason is that the impact of small distortions from x^* on aggregate welfare is negligible, while strictly softening the participation constraint of the strong country.

(iii)
$$q^{CC} > 0$$
 and $q^{SC} + q^{CC} = 1$ (Weak country always undergoing a crisis)

Even when the weak country is always undergoing a crisis, the banking union can provide risk-sharing across countries, as long as the strong country may experience a crisis with positive probability. A welfare enhancing banking union is feasible because the aggregate resolution cost associated with the provision of foreign aid to the strong country in the bank failure state (1, CC) is initially insignificant relative to the gains from the provision of foreign aid to the weak country in the bank failure state (2, SC). The underlying reason is of the same nature as in the economy described in (ii). The above examples describe the optimal banking union in the most simple economies with $q^{SC} < 1$. Observe that the main features of the resolution policies described in the stylized examples extend to the general case characterized in Proposition 3.

The proposition also states that when $q^{SC} = 1$ so that the strong country is always stable and the weak country is always undergoing a crisis, there is no possibility of creating a banking union. The reason is that in such an economy the only way of compensating the strong country for providing foreign aid in state (2, SC) would be through support in the state (1, SC). In this case, in addition to the distortion induced by the overstatement effect, there would be a cost loss in any transfer from the weak to the strong country. This is also the reason why in any optimal banking union countries undergoing crises never provide foreign aid to stable countries.

Gains and losses induced by the optimal banking union

In order to assess the gains and losses induced by the optimal banking union, we can write:

$$U_{1}\left(\widehat{\boldsymbol{x}}\right) + U_{2}\left(\widehat{\boldsymbol{x}}\right) - \left(\overline{U}_{1} + \overline{U}_{2}\right) = \underbrace{\left(1 - p\right) \left[q^{CS}\left(\Pi^{C} - \Pi(\widehat{x}_{1,CS})\right) + q^{SC}\left(\Pi^{C} - \Pi(\widehat{x}_{2,SC})\right)\right]}_{Reduction \ in \ excess \ resolution \ cost \ from \ sovereign \ crises} - \left(1 - p\right) \left[q^{SS}\left(\Pi(\widehat{x}_{1,SS}|1,SS) - \Pi^{S}\right) + q^{CC}\left(\Pi(\widehat{x}_{1,CC}|1,CC) - \Pi^{C}\right)\right].$$
(17)

Cost of excessive bail-outs at strong country

>0

>0

The first term includes the reduction of the aggregate resolution cost in the bank failure states in which there are risk-sharing motives to mutualize the funding of bail-outs. Since $\hat{x}_{1,CS} < x^* < \hat{x}_{2,SC}$, this reduction is smaller than in the ex-post optimal banking union. The second term in (17) captures the social cost of the excessive amount of bail-outs of the bank in the strong country in states in which there is no risk-sharing motive to provide foreign aid, but simply the need to compensate the strong country to form the banking union.

L

Also, observe that the participation constraint of the strong country binds, as stated in Proposition 3 (i). Hence, entering the banking union does not affect the expected aggregate welfare of the strong country's agents. However, forming a banking union leads to a redistribution of welfare in that country from the public to the bank owners, as the bank is bailed out more frequently within the banking union.

We conclude this section by highlighting a policy implication of our results: When country asymmetry is large, an aggregate welfare maximizing banking union requires the design of risk-sharing rules that are country-specific, explicitly recognizing the ex-ante differences among countries. Aiming at a symmetric banking union may render the project unfeasible. It is worth emphasizing that while the country-specific resolution policies in the optimal banking union may seem to privilege the strong country ex-post, it is indeed the weak country that appropriates the entire welfare gains associated with the creation of a banking union from an ex-ante perspective.

5 Discussion of the results

In this section we discuss the results along two additional dimensions. In Section 5.1 we analyze the sign of the expected net transfers across countries in the optimal banking union. In Section 5.2 we describe the role of initial-date unconditional transfer in restoring ex-post efficient resolution policies with large country asymmetry.

5.1 Net transfers across countries in an optimal banking union

As stated in the Introduction of the paper, a general principle governing the banking union negotiations is that it should not lead to permanent transfers across states. This position has been spearheaded by countries with strong economies in an attempt to prevent that the creation of common fiscal backstops becomes a hidden instrument to subsidize countries with weaker fiscal positions.

In this section, we argue that the optimal banking union among countries with a large degree of asymmetry characterized in Proposition 3 may lead to the weak country being a net contributor of funds. More generally, we illustrate how the degree of asymmetry and the correlation of public finances among countries shape transfers in the optimal banking union and determine which country is a net fiscal contributor.

Throughout this section we fix the value of the probability $q^{SC} > 0$. Moreover, for ease of exposition, and without any significant loss of insight, we let $q^{CC} = 0$. Notice that the assumption that $q^{CC} = 0$ implies that q^{CS} stands for the probability that the strong country undergoes a sovereign crisis. The distribution of public finances pairs at t = 1 is thus entirely described by the probability $q^{CS} \in [0, q^{SC})$, where the upper bound of the latter interval stems from Assumption 1. The value $q^{CS} = 0$ represents the case of maximal asymmetry, whereby the strong country is always stable. On the other extreme, when q^{CS} gets arbitrarily close to q^{SC} , we have that countries are (almost) symmetric. For each $q^{CS} \in [0, q^{SC})$, we denote by $T(q^{CS})$ the expected net contribution of funds from the strong country to the weak country in the optimal banking union.

Consider first the case in which the asymmetry among countries is not too large, that is, $q^{CS} > aq^{SC}$, where $a \in (0, 1)$ corresponds to the value described in Proposition 2, and thus the ex-post optimal banking union is feasible. The expected net contribution of funds by the strong country is then given by:

$$T(q^{CS}) = (1-p)\left(q^{SC} - q^{CS}\right)(1-x^*)\left(1 - G\left(\overline{k}(x^*)\right)\right) > 0.$$
(18)

When countries are not too asymmetric, the strong country is a net contributor of funds. In addition, as q^{CS} decreases—so that the strong country becomes less likely to experience a fiscal crisis—, we have that $T(q^{CS})$ increases, that is the strong country's expected net transfer increases.

This is no longer the case when the ex-post optimal banking union is not feasible, that is, when $q^{CS} < aq^{SC}$. Ensuring the participation of the strong country requires distortions of the ex-post optimal resolution policies to reduce the net transfers from the strong country to the weak country in each state. These distortions may be as large as to revert the position of the strong country as a net contributor within the banking union.

In fact, the weak country is a net contributor when the strong country is always stable, as we show below. When $q^{CS} = 0$, there are only two states in which there are transfers of funds across countries: The strong country contributes in state (2, SC) and receives a transfer in state (1, SS). Since the strong country's participation constraint is binding, it follows that the net expected transfer from the strong country must equal the increase in the aggregate resolution cost in state (1, SS) as compared to autarky, which stems from the overstatement effect that the foreign contribution to bail-outs induces without any associated cost gain. As a consequence, the strong country is a net expected receiver of funds.

The following proposition formalizes this result.

Proposition 4 (Net transfers across countries) Suppose that the strong country is always (expected to be) stable, that is, $q^{CS} = 0$. Then, the weak country is a net expected contributor to the banking union, that is, $T(q^{CS} = 0) < 0$.

Our analysis emphasizes that imposing zero net expected cross-country transfers is not the optimal way of ensuring each country's legitimate demand that the creation of a banking union is not detrimental to its constituency.

5.2 Initial date transfers

Recall from Proposition 2 that the ex-post optimal banking union is not feasible with large country heterogeneity, because it requires too large an expected net transfer from the strong to the weak country. Moreover, as shown in Proposition 3, the optimal banking union with large country heterogeneity exhibits ex-post inefficient transfers from the weak to the strong country in states in which there are no cost-of-funds gains. In this section, we explore the sources of this inefficiency by showing that the ex-post optimal banking union is achievable if we allow countries to issue unconditional transfers at the inception of the banking union.

Suppose that at t = 0 the cost of funds in the two countries is λ_S and that countries can issue transfers among them. Importantly, these funds can be employed by countries at their own will, that is, countries need not funnel these resources into the banking system. Let mdenote the amount transferred *from* the weak *to* the strong country at t = 0 (i.e., m < 0represents a case in which the strong country issues a transfer to the weak country). Then, for any given banking union \boldsymbol{x} and initial transfer m, the utility in each of the countries is given by:

$$\begin{aligned} \widehat{U}_1(m, \boldsymbol{x}) &= U_1(\boldsymbol{x}) + (1 + \lambda_S)m \\ \widehat{U}_2(m, \boldsymbol{x}) &= U_2(\boldsymbol{x}) - (1 + \lambda_S)m. \end{aligned}$$

Notice that receiving a transfer reduces the country's need to raise an amount m through distortionary taxation or public debt issuance. Hence, the value of receiving an amount m increases the country's utility by $(1 + \lambda_S)m$. Moreover, observe that transfers do not affect aggregate welfare, since the shadow cost of raising an amount m of public funds to issue the transfer is precisely $(1 + \lambda_S)m$.

A banking union \boldsymbol{x} with an associated initial transfer m is feasible if and only if no country is worse off than in autarky, that is $\widehat{U}_i(m, \boldsymbol{x}) \geq \overline{U}_i$, for i = 1, 2. The following proposition characterizes the optimal banking unions with initial date transfers.

Proposition 5 (Optimal banking union with initial date transfers) Suppose that countries can make transfers across them at the initial date t = 0. Then, there exist $m_{\min}^* < m_{\max}^*$ such that a banking union described by an initial date transfer m and resolution policies \boldsymbol{x} is optimal if and only if:

$$m \in [m^*_{\min}, m^*_{\max}]$$
 and $\boldsymbol{x} = \boldsymbol{x}^*$,

where \mathbf{x}^* is the ex-post optimal banking union characterized in Proposition 1. Moreover, $m^*_{\min} > 0$ if and only if country asymmetry is large, in the sense of Proposition 2.

The proposition states that the possibility of issuing transfers across countries at the inception of the banking union ensures that the resulting agreement features ex-post optimal resolution policies. The reason for this result is that an initial-date transfer to the strong country relaxes its participation constraint, thus avoiding the distorted contributions to bail-out funding characterized in Proposition 3. The proposition also states that there is continuum of optimal banking unions that differ only in the initial-date transfer size. This result is reminiscent of the second welfare theorem in that countries can agree on the optimal risk-sharing agreement x^* , which maximizes overall welfare, and independently issue transfers m to distribute the welfare gains from the agreement. Intuitively, when country as a compensation for its larger contribution to welfare, and thus $m^*_{\min} > 0$.

This arrangement can be interpreted as a coinsurance contract in which the weak country pays a premium m^* at the initial date to the strong country. There is coinsurance insofar as both countries commit to provide the (optimal coinsurance) amount x^* to fund bail-outs at the other country in the appropriate contingencies, as characterized in Proposition 1. The premium is paid by the weak country to compensate the strong country for its larger expected contribution, given the higher likelihood of the weak country to undergo a sovereign crisis.

Despite the benefits of allowing for unconditional transfers at the inception of the agreement, there are several reasons why this sort of arrangement may not be feasible. On the one hand, there may be domestic political economy constraints preventing fiscally weaker countries to transfer funds to strong countries in exchange for contingent help at a future date. Moreover, it may not even be feasible for countries with weaker public finances to raise a large amount of funds without compromising their fiscal positions at the beginning of the banking union.

6 Extensions

In this section we extend the model in three directions. In Section 6.1 we allow for early intervention by the central authority at an interim date and analyze how country asymmetry may lead to a differential treatment of banks across countries in these interventions. In Section 6.2 we analyze the optimal resolution policies when, in addition to direct transfers, the foreign country can issue a loan to the country undergoing a crisis. In Section 6.3 we introduce cross-border debt holdings and analyze the extent of risk-sharing in the presence of foreign ownership of debt claims.

6.1 Early intervention

So far we have circumscribed the authorities' decisions to the resolution of a bank when the realized value of its assets is not enough to repay debt investors. Authorities can in practice *early intervene* a bank to protect depositors and other senior debtholders when they believe that the bank is likely to fail at a future date. Early intervention ahead of a potential further deterioration of the bank assets allows authorities to ensure the repayment of all senior debtholders without the need to recourse to public funds.

In this section we extend the model to allow for both late resolution and early intervention actions. We analyze how these two are optimally set in a banking union. We introduce an interim date, which we refer to as t = 1/2, in which the relevant resolution authority observes a signal p_i , standing for the probability that bank *i*'s project succeeds at t = 1. At t = 0, p_i is regarded as a random variable with support [0, 1], cumulative distribution function $F(\cdot)$ and strictly positive probability density function $f(\cdot)$, with $\int_0^1 dF(p_i) = p$. We also assume that at t = 1/2 the pair of public finances $\sigma \in \{SS, SC, CS, CC\}$ at t = 1 is common knowledge. An early intervention of the bank at t = 1/2 leads to the liquidation of the bank assets, which yields a payoff of $L \ge 1$. The authority uses these proceeds to repay the unit principal owed to the bank debtholders and distributes the residual payoff among the bank's owners. Note that we assume that early intervention actions are based on a signal that is observed by both the central and domestic authorities. The rationale for this is that early intervention prevents the bail-in of debtholders, which makes the intervention process smoother and thus less reliant on information that can only be obtained by domestic authorities.

In this set-up a banking union consists of a pair of vectors $(\boldsymbol{x}, \overline{\boldsymbol{p}})$. The vector $\boldsymbol{x} = (x_{i,\sigma})$ has the same meaning as in the baseline model. The vector $\overline{\boldsymbol{p}} = (\overline{p}_{i,\sigma})$ consists of the early intervention thresholds for the interim signal, below which the relevant authority intervenes the bank.

We first analyze the ex-post optimal intervention thresholds $\overline{p}(x_{i,\sigma}|i,\sigma)$, which depend on the resolution home contributions to bail-outs $x_{i,\sigma}$ through its effect on the aggregate resolution cost. They are characterized by the following indifference condition:

$$\overline{p}(x_{i,\sigma}|i,\sigma)R - (1 - \overline{p}(x_{i,\sigma}|i,\sigma))\Pi(x_{i,\sigma}|i,\sigma) = L.$$
(19)

The left hand side of (19) accounts for the aggregate expected utility associated with letting the bank continue until t = 1 conditional on the threshold signal. The first term captures the expected return of the bank assets. The second term includes the expected aggregate resolution cost, which takes into account both the probability that the bank fails and the resolution policies that will be adopted in that contingency. The right hand side of (19) accounts for the liquidation proceeds from the early intervention of the bank.

The ex-post optimal banking union is defined as that maximizing aggregate utility in the economy. It can be easily derived by backwards induction: It consists of the ex-post optimal late resolution policies \boldsymbol{x}^* characterized in Proposition 1, and the associated ex-post optimal early intervention thresholds $\overline{\boldsymbol{p}}^* \equiv (\overline{p}(x_{i,\sigma}^*|i,\sigma))$ given by Equation (19) evaluated in each ex-post optimal value $x_{i,\sigma}^*$ of home contributions. For the same reasons as in the baseline model, the ex-post optimal banking union $(\boldsymbol{x}^*, \overline{\boldsymbol{p}}^*)$ is feasible only when country asymmetry is not large.

Let us move to the analysis of the optimal banking union when country asymmetry is large. In this case, not only late resolution decisions, but also early intervention actions, might not be ex-post optimal. We say that a banking union $(\boldsymbol{x}, \overline{\boldsymbol{p}})$ is *forbearing* in the early intervention of bank *i* in state σ if the intervention threshold $\overline{p}_{i,\sigma}$ lays below the optimal intervention threshold characterized by Equation (19), that is, if $\overline{p}_{i,\sigma} < \overline{p}(x_{i,\sigma}|i,\sigma)$. Similarly, we say that the banking union is *tight* in the early intervention of bank *i* in state σ if $\overline{p}_{i,\sigma} > \overline{p}(x_{i,\sigma}|i,\sigma)$.

In order to assess whether forbearance or tightness in early intervention are optimal we derive an analogous expression to (12) for the extended model:

$$U_{i}(\boldsymbol{x}, \overline{\boldsymbol{p}}) = \underbrace{\sum_{\sigma} q^{\sigma} \left(\left(\int_{\overline{p}_{i,\sigma}}^{1} p dF(p) \right) R + F(\overline{p}_{i,\sigma}) L \right) - 1}_{\text{NPV}} (20)$$

$$- \underbrace{\sum_{\sigma} \left(\int_{\overline{p}_{i,\sigma}}^{1} (1-p) dF(p) \right) q^{\sigma} \left(\underbrace{\prod(x_{i,\sigma}|i,\sigma)}_{\text{Aggregate resolution cost}} - \underbrace{t(x_{i,\sigma}|i,\sigma)}_{\text{Cost of transfers from foreign country}} \right)}_{\text{Home resolution cost}} - \underbrace{\sum_{\sigma} \left(\int_{\overline{p}_{i,\sigma}}^{1} (1-p) dF(p) \right) q^{\sigma} t(x_{\overline{i},\sigma}|\overline{i},\sigma) .}_{\text{Cost of transfers to foreign country}} \right)$$

Observe that, unlike (12), (20) takes into account that the bank is liquidated whenever the (state-contingent) signal lays below the (state-contingent) early intervention threshold, yielding a value of L.

Suppose that for some bank failure state (i, σ) the banking union prescribes a resolution policy given by $x_{i,\sigma} < 1$, and thus country *i* receives foreign aid in that contingency. Using (19), we have that the derivative of the expected utility in country *i* with respect to the interim threshold $\overline{p}_{i,\sigma}$, evaluated at the ex-post optimal threshold $\overline{p}_{i,\sigma} = \overline{p}(x_{i,\sigma}|i,\sigma)$, is given by:

$$\frac{\partial U_i(\boldsymbol{x},\overline{p})}{\partial \overline{p}_{i,\sigma}}\Big|_{\overline{p}_{i,\sigma}=\overline{p}(x_{i,\sigma}|i,\sigma)} = q^{\sigma} \left[\begin{array}{c} L - \overline{p}(x_{i,\sigma}|i,\sigma)R - \\ \left(1 - \overline{p}(x_{i,\sigma}|i,\sigma)\right) \left(\Pi(x_{i,\sigma}|i,\sigma) - t(x_{i,\sigma}|i,\sigma)\right) f(\overline{p}(x_{i,\sigma}|i,\sigma)) \end{array} \right], \quad (21)$$
or:

or:

$$= -q^{\sigma} t(x_{i,\sigma}|i,\sigma) f(\overline{p}(x_{i,\sigma}|i,\sigma)) < 0.$$
(22)

Hence, whenever a country receives foreign aid, early intervention forbearance increases the country's expected utility. The intuition is that foreign transfers make the bank continuation more profitable to the recipient country than to the entire economy. Moreover, the definition of the ex-post optimal intervention threshold implies that:

$$\frac{\partial W\left(\boldsymbol{x},\overline{p}\right)}{\partial \overline{p}_{i,\sigma}}\Big|_{\overline{p}_{i,\sigma}=\overline{p}\left(x_{i,\sigma}|i,\sigma\right)}=0.$$
(23)

Hence, the aggregate welfare effect of marginal changes of the early intervention threshold around its ex-post optimal value is negligible. As a result, the participation constraint of the strong country, which is binding when country asymmetry is large, can be relaxed with a negligible aggregate welfare impact by allowing some early intervention forbearance in the states in which the strong country receives foreign aid. Analogously, if the banking union prescribes some amount of foreign aid to the weak country in some state, then it is ex-ante optimal that the central authority be tight in the intervention of the weak country's bank to soften the participation constraint of the strong country. The next proposition characterizes the properties of early intervention policies in the optimal banking union:

Proposition 6 (Optimal early intervention with large country asymmetry) There exists $a' \in (0,1)$ such that the optimal banking union is ex-post optimal if and only if $q^{CS} \geq a'q^{SC}$. If $q^{CS} < a'q^{SC}$ and $q^{SC} < 1$, then in an optimal banking union $(\widehat{\boldsymbol{x}}, \widehat{p})$ the central authority early intervention policy is forbearing with the strong country's bank in states $\{SS, CS, CC\}$, and is tight in the early intervention of the weak country in state SC.

The proposition states that when country asymmetry is large the bank in the strong country will benefit from some early intervention forbearance. In contrast, the bank in the weak country will experience some tight early interventions. These two early intervention distortions complement the late resolution distortions in softening the strong country participation constraint. The preferential treatment of the bank in the strong country suggests that there could be a tension when countries differ substantially between the objective of minimizing resolution costs within a banking union and that of ensuring a level-playing field across banks.

6.2 Direct versus indirect aid

In the baseline model we have considered financial aid from the foreign country in the form of a transfer to the debtholders of the failing bank. This form of external support, which resembles the Direct Recapitalization Instrument (DRI) established by the European Stability Mechanism (ESM) in 2014, differs from the indirect recapitalization instrument (IRI), in which the ESM grants a loan to a fragile sovereign to support the resolution of domestic failing banks. The current agreements among EU members establish a pecking order in the use of these instruments: Access to the Direct Recapitalization Instrument can only be granted "if indirect recapitalization is not possible," and its use requires some contribution by the sovereign in distress.

In this section we extend the model to allow for both direct transfers and loans issued by the foreign country, and inquire about the optimality of prioritizing indirect foreign aid. We introduce an additional date, t = 2, at which the sovereigns may also raise public funds. Countries may be stable or undergo a crisis at t = 2. To keep the setup as simple as possible, we assume that a country undergoing a crisis becomes stable at t = 2 with probability $\rho_C \in (0, 1)$ and that a stable country continues to be stable with probability $\rho_S \in (0, 1]$. We denote the expected cost of public funds at t = 2 given the sovereign state $h \in \{S, C\}$ at t = 1 as $\tilde{\lambda}_h \equiv \rho_h \lambda_S + (1 - \rho_h) \lambda_C$. We restrict loans to be fiscally neutral (in expectation) for the lending country so that any subsidy among countries is channeled through direct transfers. Hence, the gross interest rate on a loan granted by a country in state $h \in \{S, C\}$ is given by $R_h = \frac{1+\lambda_h}{1+\lambda_h}$.

Formally, and similarly to the baseline model, we can describe a banking union by a pair $(x_{i,\sigma}, y_{i,\sigma})$, establishing for each of the bank failure states (i, σ) the home country contribution $x_{i,\sigma}$ and the amount $y_{i,\sigma}$ borrowed as indirect aid to fund bail-outs. The remaining amount $1 - x_{i,\sigma} - y_{i,\sigma}$ is provided by the foreign country in the form of direct foreign aid. As in the baseline model, a standard mechanism-design approach allows us to focus on resolution mechanisms in which the decision of whether to bail out or to bail in the failing bank is delegated to the home resolution authority.

In order to illustrate the role for loans in this setup, consider a bank failure while the home country is undergoing a crisis and the foreign country is stable. Partially funding a bail-out through a foreign country loan plays the same role as home contributions in reducing the willingness of the domestic authority to overstate the need for foreign support, as the loan has to be paid back at t = 2. Moreover, since the country undergoing the crisis pays back the loan at a time at which it is stable with positive probability, substituting home contributions with a loan from the foreign country reduces the expected cost of funds for the borrowing country, since $(1 + \tilde{\lambda}_C) R_S < 1 + \lambda_C$. Moreover, by construction of R_S , the loan is fiscally neutral for the lending country. Therefore, indirect foreign aid dominates home contributions. Nonetheless, since $\tilde{\lambda}_C > \lambda_S$, direct foreign aid constitutes a less costly source

of public funds from an aggregate perspective. Hence, as in the baseline model, there is an aggregate cost advantage of using direct foreign aid to fund banks' bail-outs. Consequently, the optimal funding mix consists of a combination of a direct transfer and a loan from the foreign country, thereby eliminating home contributions in the resolution of banks.

When both countries are either concurrently stable, or simultaneously undergoing a crisis, there is no gain from using loans. Hence, any loan could be substituted out by a contemporaneous home contribution with no payoff consequence for any of the countries. In order to simplify the exposition, in the following proposition we focus on the set of mechanisms in which the use of loans strictly improves welfare. A similar analysis to that conducted in Section 4 leads to the following result:

Proposition 7 (Banking union with direct and indirect foreign aid) Assume that $q^{SC} < 1$. The optimal banking union (\hat{x}, \hat{y}) is such that:

(i) In states $(i, \sigma) \in \{(1, CS), (2, SC)\}$ bail-outs are funded with a combination of strictly positive amounts of direct (transfer) and indirect (loan) foreign aid-and no contemporaneous home contributions-, that is, $\hat{x}_{i,\sigma} = 0$ and $\hat{y}_{i,\sigma} \in (0,1)$ (and thus $1 - \hat{x}_{i,\sigma} - \hat{y}_{i,\sigma} \in (0,1)$).

(ii) In states $(i, \sigma) \notin \{(1, CS), (2, SC)\}$, indirect (loan) foreign aid is not issued, that is, $\hat{y}_{i,\sigma} = 0$, and thus any source of foreign aid is channeled through direct transfers.

The proposition states that in every contingency in which it is optimal to issue loans, these must be combined with a direct transfer from the issuing country to reduce the aggregate cost of bail-outs. In addition, in those states, it is also optimal not to require home contributions, as the loans provide incentives for local authorities not to overstate the need for bail-outs.

Unlike the ESM design of the DRI as a last resort instrument, this result suggests that the DRI should be considered as an integral part of optimal resolution schemes, which should be systematically used in combination with the IRI. Moreover, the access to the DRI should be granted without requiring the home sovereign undergoing a crisis to contribute to the funding of a bail-out contemporaneously.

6.3 Cross-border debt holdings

In this section we show that the presence of cross-border liabilities provides an additional motive for countries to form a banking union and increases the welfare gains generated by such an agreement. Formally, we assume that a fraction β of each bank's debt is held by agents of the foreign country. The case $\beta = 0$ thus corresponds to the baseline model. We keep the assumption that bail-ins originate a disruption cost of k at the home country that is privately observed by the home authority. This assumption is particularly plausible for banks whose core lending activities are domestic, in which case the potential contagion to other banks will be presumably confined to domestic institutions whose assets are more correlated with those of the failing bank.²⁶

Consider the resolution of a bank at a country undergoing a crisis when the foreign country is stable. Let x denote the home contribution to a bail-out in the baseline economy and x' a home contribution in the economy with cross-border liabilities. The following comparison ensues:

Utility following bail-out	Cross-border	Baseline
Home	$1 - \beta - (1 + \lambda_C)x'$	$1 - (1 + \lambda_C)x$
Foreign	$\beta - (1 + \lambda_S)(1 - x')$	$-(1+\lambda_S)(1-x)$

Setting $x' \equiv x - \frac{\beta}{1+\lambda_C}$, it follows that the utility-upon-bail-out difference between the cross-border and the baseline economy is given by:

Home	$1 - \beta - (1 + \lambda_C)x' - (1 - (1 + \lambda_C)x) = 0$
Foreign	$\beta - (1 + \lambda_S)(1 - x') - (-(1 + \lambda_S)(1 - x)) = \frac{\lambda_C - \lambda_S}{1 + \lambda_C} > 0$

These expressions show that starting with any home contribution x to bail-outs in the baseline model, we can reduce the home contribution to x' in the case in which there are crossborder debt holdings, maintaining the home country's utility unaltered and strictly increasing the foreign country's utility. The reason is that when a fraction of the bank's debt is held abroad, the foreign country's contribution to a bail-out constitutes a partial transfer to its own agents.

These arguments can be generalized to the remaining bank-failure states, as we state in the following proposition.

Proposition 8 (Banking union with cross-border liabilities) Suppose that a fraction $\beta > 0$ of each bank's debt is held by agents from the foreign country. Then the welfare

 $^{^{26}}$ Under the assumption that a bail-in generates contagion costs also in the foreign country, and that its magnitude is privately observed by the foreign authority, the mechanism design problem would be such that the principal (the central authority) has to provide incentives to reveal their private information to two agents-both the home and the foreign authority. The formal analysis in this case becomes much more complex and is beyond the scope of this paper.

gains relative to autarky induced by the optimal banking union are strictly larger than in the baseline model.

The presence of cross-border liabilities provides an additional motive for countries to be willing to share the costs of bailing out banks. The proposition shows that such motive adds to the risk-sharing motive highlighted in the baseline model and increases the welfare gains that the optimal banking union attains.

7 Conclusion

In this paper, we analyze the optimal bank resolution and the design of a fiscal backstop in a banking union. The rationale for a banking union is to provide a risk-sharing agreement across countries when one of the sovereigns may be in financial distress at the time of having to resolve a failing domestic bank. We address the potential for risk-sharing in a banking union in a setup that takes into consideration two features that are prevalent in the eurozone banking union project: Supranational resolution authorities must rely on the information provided by domestic resolution authorities and countries differ on the strength of their public finances.

The informational friction limits the extent of risk-sharing because countries face an incentive to overstate the bail-in cost to obtain funding from the neighboring country to bail out home troubled banks. The ex-post aggregate welfare resolution policies in a banking union entail co-funding of bail-outs if and only if banks are located in countries undergoing a crisis and the foreign country is stable. In addition, ex-post optimal decisions have to be distorted when country asymmetry is large to ensure the participation of the strong country in the banking union. In particular, we find that the required distortions reduce the amount of foreign aid to the weak country and increase the transfers to the strong country, thereby increasing the probability of bail-outs in the strong country and decreasing it in the weak country.

The findings in the paper highlight the importance of establishing country-specific resolution policies for heterogenous countries to benefit from a banking union. However, even though asymmetric resolution policies privilege the strong country's bank, it is the weak country's agents that appropriate the surplus created by the banking union. Indeed, the optimal banking union when country asymmetry is large leads to a net welfare gain of the weak country's agents, and to a redistribution of welfare from the taxpayer to the bank owners within the strong country.

The paper also highlights that requiring expected transfers to net out across countries is suboptimal. In particular, we find that strong countries should be the net receiver of funds when their public finances are always stable.

The need for country-specific policies to ensure the participation of the strong countries in the union extends to the case in which authorities can early intervene a bank at a time in which the liquidation of the bank assets yields sufficient proceeds to repay debtholders in full. We find that the optimal banking union exhibits forbearance in the early intervention decisions of the bank in the strong country and is tight in intervening the weak country's bank.

Our results also have implications for the design of the European Stability Mechanism assistance programs. We find that the current "pecking order", which envisions the direct recapitalization instrument as a last recourse option, is suboptimal: Combining loans with direct transfers can improve upon the existing regulation.

Finally, the paper highlights a risk-sharing rationale for a banking union. Nonetheless, the presence of cross-country debt exposures among banking sectors facilitates risk-sharing agreements and enhances the welfare that countries can attain through a banking union.

Appendix

A Truth-telling bank resolution mechanisms

In this appendix we first characterize the different types of truth-telling resolution mechanisms and then select the only type that is relevant for the analysis of an optimal banking union. We show, as stated in Section 2, that the relevant truth-telling mechanisms can be described by the required home contribution to bail-outs x, and can be implemented by delegating the resolution decision (bail-out or bail-in) to the home authority.

We first adopt without loss of generality the following convention: for the knife-edge cases in which under a particular resolution mechanism the home authority is indifferent between reporting a value of k that leads to a bail-out and another one leading to a bail-in, we assume that it chooses to report the former.²⁷

Let $\{b_{i,\sigma}(\hat{k}), x_{i,\sigma}(\hat{k})\}\$ be a resolution mechanism in bank failure state (i, σ) . Suppose that the bail-in cost observed by the home authority is k. The expected utility of agents in country i if the home authority reports \hat{k} is given by:

$$V(\widehat{k},k) = \underbrace{b_{i,\sigma}(\widehat{k})}_{Debtholders' payoff} - \underbrace{(1 - b_{i,\sigma}\left(\widehat{k}\right))k}_{Bail-in \ Cost} - \underbrace{(1 + \lambda_{i,\sigma}) x_{i,\sigma}(\widehat{k})}_{Cost \ Public \ Funds}, \tag{A1}$$

where $\lambda_{i,\sigma}$ denotes the deadweight cost associated with raising public funds in country *i* in the state of public finances σ . The first term in (A1) corresponds to the transfer that the bank's debtholders receive through a bail-out. The second term stands for the bail-in costs. The last term includes the expected gross social cost of the funds country *i* has to provide for the resolution of its bank. The truth-telling property of the resolution mechanism can then be written as:

$$V(k,k) \ge V(\widehat{k},k)$$
 for all k and $\widehat{k} \in [0, k_{\max}].$ (A2)

The next lemma characterizes the set of incentive compatible resolution mechanisms.

Lemma 1 A resolution mechanism for a failing bank in a country is truth-telling if and only if it belongs to one of the three following classes:

- 1. (Unconditional bail-in) $b_{i,\sigma}(k) = 0$ for all $k \in [0, k_{\max}]$.
- 2. (Unconditional bail-out) There exists $\bar{x} \in [0,1]$ such that $b_{i,\sigma}(k) = 1$ and $x_{i,\sigma}(k) = \bar{x}$ for all $k \in [0, k_{\max}]$.

 $^{^{27}}$ Without this convention, there could be other truth-telling schemes on top of those described in the ensuing Lemma 1 in which the close interval is open on its left boundary. Nonetheless, these other mechanisms are ex-ante payoff equivalent to those described in Lemma 1.

3. (Threshold) There exists $x \in (\underline{x}, 1]$ and $\overline{k}_{i,\sigma}(x) \in (0, k_{\max})$ such that $b_{i,\sigma}(k) = 1$ if and only if $k \ge \overline{k}_{i,\sigma}(x)$, in which case $x_{i,\sigma}(k) = x$. Moreover, we have that

$$\overline{k}_{i,\sigma}(x) = (1 + \lambda_{i,\sigma})x - 1, \tag{A3}$$

where $\underline{x} \in (0,1)$ is uniquely determined by the condition $\overline{k}_{i,\sigma}(\underline{x}) = 0$.

Proof Necessity Suppose that $\left\{b_{i,\sigma}\left(\hat{k}\right), x_{i,\sigma}\left(\hat{k}\right)\right\}$ is an incentive compatible (truth-revealing) mechanism. Let us define the set $\Gamma^1 = \left\{k \in [0, k_{\max}] \text{ s.t. } b_{i,\sigma}(k) = 1\right\}$. If Γ^1 is empty then case 1 in the lemma is satisfied. Let us suppose that Γ^1 is non empty. Taking into account that $(1 + \lambda_{i,\sigma})x$ is strictly increasing in x, it is immediate to show that truth-telling implies that there exists a constant \overline{x} such that $x_{i,\sigma}(k) = \overline{x}$ for all $k \in \Gamma^1$.

Let us now prove that if $k' \geq k$ for some $k \in \Gamma^1$ then $k' \in \Gamma^1$. In fact, incentive compatibility implies

$$V(k',k') \ge V(k,k') \Leftrightarrow \begin{array}{l} b_{i,\sigma}(k') - (1 - b_{i,\sigma}(k'))k' - (1 + \lambda_{i,\sigma})x_{i,\sigma}(k') \ge \\ \ge 1 - (1 + \lambda_{i,\sigma})\overline{x}, \\ V(k,k) \ge V(k',k) \Leftrightarrow \begin{array}{l} 1 - (1 + \lambda_{i,\sigma})\overline{x} \ge \\ \ge b_{i,\sigma}(k') - (1 - b_{i,\sigma}(k'))k - (1 + \lambda_{i,\sigma})x_{i,\sigma}(k'), \end{array}$$

and combining the two inequalities we have

$$-(1 - b_{i,\sigma}(k'))k' \ge -(1 - b_{i,\sigma}(k'))k,$$

which can only be satisfied if $b_{i,\sigma}(k') = 1$.

Let $\overline{k} = \inf(\Gamma^1)$. We must have $\overline{k} \in \Gamma^1$. In fact, if that is not the case then $b_{i,\sigma}(\overline{k}) = 0$. By definition of \overline{k} , for any $\varepsilon > 0$ there exists $k' \in \Gamma^1$ and such that $0 < k' - \overline{k} < \varepsilon$. Incentive compatibility implies:

$$V(\overline{k},\overline{k}) \geq V(k',\overline{k}) \Leftrightarrow -\overline{k} \geq 1 - (1 + \lambda_{i,\sigma})\overline{x}$$
$$V(k',k') \geq V(\overline{k},k') \Leftrightarrow 1 - (1 + \lambda_{i,\sigma})\overline{x} \geq -k'.$$

Combining the two inequalities and taking limits $\varepsilon \to 0$ we conclude that

$$-\overline{k} = 1 - (1 + \lambda_{i,\sigma})\,\overline{x}.$$

But then when the local authority observes \overline{k} it is indifferent between reporting \overline{k} , which leads to a bail-in, and reporting any $k \in \Gamma^1$, which leads to a bail-out. Our convention established at the beginning of this Appendix then states that the local authority would report $k \in \Gamma^1$ and the mechanism would not be truth-revealing. Hence we must have $\overline{k} \in \Gamma^1$ and the fact that for all $k \geq \overline{k}$ we have $k \in \Gamma^1$ then implies that $\Gamma^1 = [\overline{k}, 1]$. If $\overline{k} = 0$ then case 2 in the lemma is satisfied. If $\overline{k} > 0$ then for any $k < \overline{k}$ it has to be the case that $b_{i,\sigma}(k) = 0$ and incentive compatibility implies

$$V(\overline{k},\overline{k}) \geq V(k,\overline{k}) \Leftrightarrow 1 - (1 + \lambda_{i,\sigma}) \overline{x} \geq -\overline{k}$$
$$V(k,k) \geq V(\overline{k},k) \Leftrightarrow -k \geq 1 - (1 + \lambda_{i,\sigma}) \overline{x}.$$

Taking limits $k \to \overline{k}$ and combining the two inequalities we have that

$$1 - (1 + \lambda_{i,\sigma})\overline{x} = -\overline{k},$$

and then case 3 in the lemma is satisfied.

Sufficiency It is straightforward to check that the mechanisms in the 3 cases in the lemma satisfy incentive compatibility. \blacksquare

Some of the truth telling mechanisms characterized in the previous lemma cannot be part of an optimal banking union because they are Pareto dominated. We have:

Lemma 2 The following truth-telling mechanisms are Pareto dominated and can be discarded for the analysis of the ex-post optimal resolution policies and of the optimal banking union:

- 1. The unconditional bail-in mechanism is Pareto dominated by a threshold mechanism with home contribution to bail-outs x = 1.
- 2. The unconditional bail-out mechanism characterized by a home contribution $\bar{x} > \underline{x}$ to bail-outs is Pareto dominated by a threshold mechanism with home contribution to bail-outs $x = \bar{x}$.

Proof The first statement is an immediate consequence of the assumption $c_{\text{max}} > \lambda_C$ and the fact that the foreign country is indifferent between an unconditional bail-in policy and a threshold mechanism in which bail-outs are entirely locally funded (i.e., such that x = 1).

Second, by construction the home country prefers a threshold mechanism with home contribution to bail-outs $x = \bar{x}$ to an unconditional bail-out mechanism with the same home contribution. The foreign country also prefers the former because it strictly reduces the probability that it has to contribute to fund a bail-out.

We conclude that it suffices to restrict our attention to the threshold mechanisms, which are described by a home contribution to bail-outs $x > \underline{x}$, and the unconditional bail-out mechanisms with home contribution to bail-outs $\overline{x} \leq \underline{x}$, which can be also thought as a family of threshold mechanisms where the induced bail-out threshold is $\overline{k} = 0$. This justifies our restriction in the entire paper to resolution mechanisms that are described by the required home contribution to bail-outs x in the entire interval $x \in [0, 1]$, and in which the decision on whether to conduct a bail-out or a bail-in is delegated to the home authority.

B Proofs

Proof of Proposition 1 We have proved all the statements of the proposition in the text preceding the proposition, except for:

i) No $x \leq \frac{1}{1+\lambda_C}$ is optimal.

In fact, we have that:

$$\Pi(1) = \int_0^{\lambda_C} k dG(k) + \lambda_C (1 - G(\lambda_C)) < \int_0^{k_{\max}} k dG(k) < \lambda_S = \Pi(0),$$

where the inequality follows from Assumption 2. Since $\Pi'(x) > 0$ for all $x < \frac{1}{1+\lambda_C}$, the inequality above implies that $\Pi(x) \ge \Pi(0) > \Pi(1)$ for all $x \le \frac{1}{1+\lambda_C}$.

ii) The function $\Pi(x)$ is convex in the interval $x \in \left[\frac{1}{1+\lambda_C}, 1\right]$ and thus has a unique minimum in this interval.

Plugging the expression for the p.d.f. and the c.d.f. of a uniform distribution in the expression for $\Pi'(x)$ in (6) we have that:

$$\Pi'(x) = (\lambda_C - \lambda_S) \frac{k_{\max} - \overline{k}(x)}{k_{\max}} - (1 + \lambda_S) (1 - x) (1 + \lambda_C) \frac{1}{k_{\max}}.$$

Differentiating the latter expression with respect to x we obtain:

$$\Pi''(x) = \frac{1+\lambda_C}{k_{\max}} \left[-(\lambda_C - \lambda_S) + (1+\lambda_S) \right] = \frac{1+\lambda_C}{k_{\max}} \left(1 + 2\lambda_S - \lambda_C \right) > 0,$$

where in the last inequality we have used that Assumption 2 implies that $2\lambda_S > k_{\text{max}} > \lambda_C$.

Proof of Proposition 2 The expression for the gains from entering the ex-post optimal banking union in (8) can be easily obtained from the general expression for the utility of agents in a country within any banking union in (12).

From (8) we have that $U_1(\boldsymbol{x}^*) - \overline{U}_1 \ge 0$ if and only if:

where:

$$a \equiv \frac{(1-x^*)(1+\lambda_S)\left(1-G\left(k(x^*)\right)\right)}{\Pi^C - \Pi(x^*) + (1-x^*)(1+\lambda_S)\left(1-G\left(\bar{k}(x^*)\right)\right)} \in (0,1).$$

Finally, from Assumption 1, and using the fact that $q^{SC} > q^{CS}$, we have that if the participation constraint of country 1 is satisfied then the participation constraint of country 2 is also satisfied. We conclude that a banking union with ex-post optimal resolution policies satisfies the participation constraint of the two countries if and only if (B1) is satisfied.

Proof of Proposition 3 We refer to a general banking union as x. Let us first state some properties of the partial derivatives of the aggregate welfare function and the expected utility in each country that have either been discussed in Section 4.3 or that can be easily derived from the expressions introduced in that section.

• For any (i, σ) :

$$\frac{\partial W\left(\boldsymbol{x}\right)}{\partial x_{i,\sigma}} = -(1-p)q^{\sigma}\frac{\partial \Pi(x_{i,\sigma}|i,\sigma)}{\partial x_{i,\sigma}}.$$

• For any i and $\sigma \in \{SS, CC\}$, if $q^{\sigma} > 0$:

$$\frac{\partial W\left(\boldsymbol{x}\right)}{\partial x_{i,\sigma}} > 0 \text{ if } x_{i,\sigma} < 1, \text{ and } \frac{\partial W\left(\boldsymbol{x}\right)}{\partial x_{i,\sigma}} = 0 \text{ if } x_{i,\sigma} = 1.$$

• For $(i, \sigma) \in \{(1, SC), (2, CS)\}$, if $q^{\sigma} > 0$: $\partial W($

$$\frac{\partial W\left(\boldsymbol{x}\right)}{\partial x_{i,\sigma}} > 0$$

• For
$$(i, \sigma) \in \{(1, CS, (2, SC))\}$$
, if $q^{\sigma} > 0$:
 $\frac{\partial W(\boldsymbol{x})}{\partial x_{i,\sigma}} > 0$ if $\frac{1}{1 + \lambda_{i,\sigma}} < x_{i,\sigma} < x^*$, $\frac{\partial W(\boldsymbol{x})}{\partial x_{i,\sigma}} = 0$ if $x_{i,\sigma} = x^*$, and $\frac{\partial W(\boldsymbol{x})}{\partial x_{i,\sigma}} < 0$ otherwise.

• For any (i, σ) , if $q^{\sigma} > 0$:

$$\frac{\partial U_{i}\left(\boldsymbol{x}\right)}{\partial x_{i,\sigma}} < 0, \text{ and } \frac{\partial U_{i}\left(\boldsymbol{x}\right)}{\partial x_{\bar{i},\sigma}} > 0.$$

Let $\hat{\boldsymbol{x}} \equiv (\hat{x}_{i,\sigma})$ be an optimal banking union. Let us denote PC_i the participation constraint of country *i*. We first prove the existence of a feasible banking union when $q^{SC} < 1$ and the properties of the optimal one. We then prove that there is no proper feasible banking union when $q^{SC} = 1$.

Optimal banking union when $q^{SC} < 1$

We conduct the proof of the statements in the proposition in a sequence of steps.

i) Any optimal banking union strictly increases aggregate welfare relative to autarky.

It suffices to find a banking union that strictly increases aggregate welfare relative to autarky. Since $q^{SC} < 1$, at least one of the three following inequalities is satisfied: $a) \ a^{CS} > 0$

From (12) we have that

$$\frac{1}{q^{CS}} \frac{\partial U_1(\mathbf{1})}{\partial x_{1,CS}} = \frac{1}{q^{SC}} \frac{\partial U_2(\mathbf{1})}{\partial x_{2,SC}} = -(1-p)(1+\lambda_C) \left(1-G\left(\lambda_C\right)\right), \quad (B2)$$

$$\frac{1}{q^{CS}} \frac{\partial U_2(\mathbf{1})}{\partial x_{1,CS}} = \frac{1}{q^{SC}} \frac{\partial U_1(\mathbf{1})}{\partial x_{2,SC}} = (1-p)(1+\lambda_S) \left(1-G\left(\lambda_C\right)\right),$$

where **1** denotes the autarky banking union. Let us consider the family of banking unions $\boldsymbol{x}(y)$, indexed by the parameter $y \geq 0$, which is characterized by: $x_{1,CS}(y) = 1 - y, x_{2,SC}(y) = 1 - \gamma y$ with $\gamma \equiv \frac{q^{CS}(1+\lambda_C)}{q^{SC}(1+\lambda_S)} - \varepsilon$ and $\varepsilon > 0$, and $x_{i,\sigma}(y) = 1$ for any other (i, σ) . Using (B2), we have that

$$\begin{aligned} \frac{d\widehat{U}_{1}(\boldsymbol{x}(y))}{dy} \bigg|_{y=0} &= -\frac{\partial U_{1}(1)}{\partial x_{1,CS}} - \gamma \frac{\partial U_{1}(1)}{\partial x_{2,SC}} = \\ &= \varepsilon (1-p)q^{SC}(1+\lambda_{S}) \left(1-G\left(\lambda_{C}\right)\right), \text{ and} \\ \frac{d\widehat{U}_{2}(\boldsymbol{x}(y))}{dy} \bigg|_{y=0} &= -\frac{\partial U_{2}(1)}{\partial x_{1,CS}} - \gamma \frac{\partial U_{2}(1)}{\partial x_{2,SC}} = \\ &= \left(1-p\right) \left(1-G\left(\lambda_{C}\right)\right) \left[\frac{(1+\lambda_{C})^{2}-(1+\lambda_{S})^{2}}{1+\lambda_{S}}q^{CS} - \varepsilon q^{SC}\left(1+\lambda_{C}\right)\right]. \end{aligned}$$

We have that since $q^{CS} > 0$, then for ε sufficiently small $\frac{d\hat{U}_1(\boldsymbol{x}(y))}{dy}\Big|_{y=0} > 0$ and $\frac{d\hat{U}_1(\boldsymbol{x}(y))}{dy}\Big|_{y=0} > 0$. Then, there exists y > 0 such that in the banking union $\boldsymbol{x}(y)$ both countries are strictly better off than in autarky.

b) $q^{SS} > 0$.

From (12) we have that

$$\frac{\partial U_1(\mathbf{1})}{\partial x_{1,SS}} = -\frac{\partial U_2(\mathbf{1})}{\partial x_{1,SS}} = -(1-p)q^{SS}(1+\lambda_S)\left(1-G\left(\lambda_S\right)\right). \tag{B3}$$

Let us consider the family of banking unions $\boldsymbol{x}(y)$ indexed by the parameter $y \geq 0$ characterized by: $x_{1,SS}(y) = 1 - y, x_{2,SC}(y) = 1 - \gamma y$ with $\gamma \equiv \frac{q^{SS}(1-G(\lambda_S))}{q^{SC}(1-G(\lambda_C))} - \varepsilon$ and $\varepsilon > 0$, and $x_{i,\sigma}(y) = 1$ for any other (i, σ) . Using (B2) and (B3), the arguments of the previous case can be reproduced in an analogous manner and show that there exists y > 0 such that in the banking union $\boldsymbol{x}(y)$ both countries are strictly better off than in autarky.

c)
$$q^{CC} > 0$$
.

Analogous to case b.

Notice that if $q^{\sigma} = 0$ for some σ , the specific resolution policies in the bank failure states (i, σ) are irrelevant. For the sake of shortening our arguments we will assume throughout the rest of the proof of the case $q^{SC} < 1$ that $q^{\sigma} > 0$ for all $\sigma \in \{SS, CC, CS\}$. The reader can easily check that our arguments are valid if $q^{\sigma} > 0$ for some $\sigma \in \{SS, CC, CS\}$, which has to be the case since $q^{SC} < 1$.

ii) Exactly one of the participation constraints is binding.

Suppose none of the participation constraints is binding. Then $\hat{x} = x^*$. But we have $q^{CS} < aq^{SC}$, and Proposition 2 implies x^* is not feasible.

Suppose PC_i is binding. Then $PC_{\overline{i}}$ is not binding because otherwise i) would imply that \hat{x} is not optimal.

iii) If PC_1 is binding then \hat{x} satisfies $\hat{x}_{1,SS} < \hat{x}_{2,SS} = 1, \hat{x}_{1,CC} < \hat{x}_{2,CC} = 1, \hat{x}_{2,CS} = 1, \hat{x}_{1,CS} < x^*$.

If PC_1 is binding then *ii*) implies that PC_2 is not binding.

For $\sigma \neq SC$, the properties of the derivatives of $W(\boldsymbol{x})$ and $U_1(\boldsymbol{x})$ with respect to $x_{2,\sigma}$ when $x_{2,\sigma} < 1$ and the optimality of $\hat{\boldsymbol{x}}$, immediately imply that $\hat{x}_{2,SS} = \hat{x}_{2,CC} = \hat{x}_{2,CS} = 1$.

For $\sigma \in \{SS, CC\}$, the properties of the derivatives of $W(\boldsymbol{x})$ and $U_1(\boldsymbol{x})$ with respect to $x_{1,\sigma}$ when $x_{1,\sigma} = 1$ and the optimality of $\hat{\boldsymbol{x}}$, immediately imply that $\hat{x}_{1,SS} < 1$ and $\hat{x}_{1,CC} < 1$.

The properties of the derivatives of $W(\boldsymbol{x})$ and $U_1(\boldsymbol{x})$ with respect to $x_{1,CS}$ when $x_{1,CS} \geq x^*$ and the optimality of $\hat{\boldsymbol{x}}$, immediately imply that $\hat{x}_{1,CS} < x^*$.

iv) If PC_1 is binding then \hat{x} satisfies $x^* < \hat{x}_{2,SC} < 1$.

If PC_1 is binding then *ii*) implies that PC_2 is not binding and the properties of \hat{x} in *iii*) are satisfied.

The properties of the derivatives of $W(\boldsymbol{x})$ and $U_1(\boldsymbol{x})$ with respect to $x_{2,SC}$ when $x_{2,SC} \in \left[\frac{1}{1+\lambda_C}, x^*\right]$ and the optimality of $\hat{\boldsymbol{x}}$, imply that $\hat{x}_{2,SC} \notin \left[\frac{1}{1+\lambda_C}, x^*\right]$. Suppose that $\hat{x}_{2,SC} < \frac{1}{1+\lambda_C}$ and thus $\overline{k}(\hat{x}_{2,SC}|2,SC) = 0$. We thus have that:

$$\Pi(\hat{x}_{2,SC}|2,SC) \ge \Pi(0|2,SC) > \Pi(1|2,SC), \tag{B4}$$

where the last inequality follows from Assumption 2.

Let us now prove that the following inequality holds:

$$\Pi(\hat{x}_{1,CS}|1,CS) > \Pi(1|1,CS).$$
(B5)

We have from *iii*) that $\hat{x}_{1,CS} < x^*$. If $\hat{x}_{1,CS} \leq \frac{1}{1+\lambda_C}$ the same arguments as for the bank failure state (2, SC) imply the inequality. Suppose that $\hat{x}_{1,CS} > \frac{1}{1+\lambda_C}$ so that $\overline{k}(\hat{x}_{1,CS}|1, CS) > 0$. Let us denote $\mu > 0$ the Lagrange multiplier of the participation constraint of country 1 at the banking union \hat{x} . Optimality of \hat{x} implies that

$$\frac{\partial W}{\partial x_{2,SC}}\Big|_{x_{2,SC}=\widehat{x}_{2,SC}} \ge -\mu \left. \frac{\partial U_1}{\partial x_{2,SC}} \right|_{x_{2,SC}=\widehat{x}_{2,SC}}, \text{ with equality iff } \widehat{x}_{2,SC} > 0.$$
(B6)

Using that $\overline{k}(x_{2,SC}|2,SC) = 0$ for all $x_{2,SC} < \frac{1}{1+\lambda_C}$, we can compute the marginal derivatives in the inequality above and obtain after some algebra

$$\mu \ge \frac{\lambda_C - \lambda_S}{1 + \lambda_S}.\tag{B7}$$

In addition, using that $\hat{x}_{1,CS} \in \left(\frac{1}{1+\lambda_C}, x^*\right)$ the optimality of \hat{x} implies that:

$$\frac{\partial W}{\partial x_{1,CS}}\Big|_{x_{1,CS}=\widehat{x}_{1,CS}} = -\mu \left. \frac{\partial U_1}{\partial x_{1,CS}} \right|_{x_{1,CS}=\widehat{x}_{1,CS}},\tag{B8}$$

which after some algebra can be written as:

$$[\mu (1 + \lambda_C) + \lambda_C - \lambda_S] = \frac{(1 + \lambda_S) (1 - \hat{x}_{1,CS}) (1 + \lambda_C)}{k_{\max} - ((1 + \lambda_C) \hat{x}_{1,CS} - 1)}.$$
(B9)

Notice that x^* is defined as the solution of (B8) for $\mu = 0$, and thus from the equality above the following equality follows:

$$\lambda_C - \lambda_S = \frac{(1+\lambda_S)(1-x^*)(1+\lambda_C)}{(k_{\max} - ((1+\lambda_C)x^* - 1))}.$$
(B10)

Dividing (B9) by (B10) we obtain:

$$\frac{1+\lambda_C}{\lambda_C - \lambda_S}\mu + 1 = \frac{1-\widehat{x}_{1,CS}}{1-x^*} \frac{k_{\max} - ((1+\lambda_C)x^* - 1)}{k_{\max} - ((1+\lambda_C)\widehat{x}_{1,CS} - 1)}.$$

Using (B7) an recognizing that $\hat{x}_{1,CS} < x^*$, it follows from the latter expression that:

$$2 < \frac{1+\lambda_C}{1+\lambda_S} + 1 \le \frac{1+\lambda_C}{\lambda_C - \lambda_S} \mu + 1 = \\ = \frac{1-\hat{x}_{1,CS}}{1-x^*} \frac{k_{\max} - ((1+\lambda_C)x^* - 1)}{k_{\max} - ((1+\lambda_C)\hat{x}_{1,CS} - 1)} < \frac{1-\hat{x}_{1,CS}}{1-x^*},$$

and thus:

$$1 - x^* < x^* - \hat{x}_{1,CS}.$$

Finally, since the function $\Pi(x_{1,CS}|1, CS)$ achieves a minimum at $x_{1,CS} = x^*$, the inequality above implies (B5).

For any other $(i, \sigma) \notin \{(1, CS), (2, SC)\}$, it follows from *iii*) and the properties of the derivatives of $W(\mathbf{x})$ with respect to $x_{i,\sigma}$ that:

$$\Pi(\widehat{x}_{i,\sigma}|i,\sigma) \ge \Pi(1|i,\sigma). \tag{B11}$$

Taking (B4), (B5) and (B11) together we have that $W(\hat{x}) < W(1)$, which contradicts the optimality of \hat{x} . We conclude that it cannot be the case that $\hat{x}_{2,SC} < \frac{1}{1+\lambda_C}$, and thus we must have $\hat{x}_{2,SC} > x^*$.

Finally, taking *iii*) into consideration, if $\hat{x}_{2,SC} = 1$ it would follow that all the home contributions of country 2 are equal to 1. Moreover, since $q^{SC} < 1$, there would be at least one home contribution of country 1 that occurs with positive probability and is different from 1. As a result, PC_2 would not be satisfied.

v) If PC_1 is binding then \hat{x} satisfies $\hat{x}_{1,SC} = 1$.

Suppose that PC_1 . Let $\mu > 0$ denote again the Lagrange multiplier of the participation constraint of country 1 at \hat{x} . Using *iv*), we have that (B6) is satisfied with equality. Some algebra rearrangements lead to:

$$\frac{\lambda_C - \lambda_S - \mu (1 + \lambda_S)}{1 + \mu} = \frac{(1 + \lambda_S) (1 - \hat{x}_{2,SC}) (1 + \lambda_C)}{k_{\max} - \bar{k}(\hat{x}_{2,SC}|2,SC)} > 0,$$
 (B12)

and thus the Lagrangian multiplier μ satisfies:

$$\mu < \frac{\lambda_C - \lambda_S}{1 + \lambda_S}.\tag{B13}$$

Suppose that $\hat{x}_{1,SC} < 1$. The optimality of \hat{x} implies that:

$$\frac{\partial W}{\partial x_{1,SC}}\Big|_{x_{1,SC}=\hat{x}_{1,SC}} \ge -\mu \left. \frac{\partial U_1}{\partial x_{1,SC}} \right|_{x_{1,SC}=\hat{x}_{1,SC}} \text{ with equality iff } \hat{x}_{1,SC} > 0.$$
(B14)

Let us first consider the case that $\hat{x}_{1,SC} > \frac{1}{1+\lambda_S}$ so that (B14) is satisfied with equality and $\bar{k}(\hat{x}_{1,SC}|1,SC) > 0$. Some algebra rearrangements lead to:

$$\mu(1+\lambda_S) - (\lambda_C - \lambda_S) = \frac{(1+\lambda_C)\left(1-\widehat{x}_{1,SC}\right)\left(1+\lambda_S\right)}{k_{\max} - \overline{k}(\widehat{x}_{1,SC}|1,SC)} > 0,$$

which implies that (B13) cannot be satisfied. Suppose, on the contrary, that $\hat{x}_{1,SC} \leq \frac{1}{1+\lambda_S}$, and thus $\bar{k}(\hat{x}_{1,SC}|1,SC) = 0$. We have that (B14) can be written as:

$$\mu \leq \frac{\lambda_C - \lambda_S}{1 + \lambda_S}$$
 with equality iff $\widehat{x}_{1,SC} > 0$,

which, taking into consideration (B13), implies that $\hat{x}_{1,SC} = 0$. Let us now show that we would have $U_1(\hat{x}) > U_1(1)$, thereby contradicting the fact that PC_1 is binding. In fact, taking into account *iii*) we have that an optimal banking union with PC_1 binding strictly increases the utility in country 1 conditional on the pair of public finances at t = 1 being $\sigma \neq SC$. Moreover, taking into account that $\hat{x}_{1,SC} = 0$, the utility in country 1 conditional on the pair of public finances at t = 1 being $\sigma = SC$ is given by:

$$U_1(\widehat{\boldsymbol{x}}|\sigma = SC) = pR + (1-p) \cdot 1 - (1-p)(1-\widehat{x}_{2,SC})(1+\lambda_S) \left(1 - G\left(\overline{k}(\widehat{x}_{2,SC}|2,SC)\right)\right), \quad (B15)$$

where the first term captures the expected payoff of the bank project, the second term includes the transfers to country 1 induced by the bail-outs entirely funded by country 2 whenever bank 1 fails, and the last term captures the transfers country 1 does to country 2 to contribute to the funding of bail-outs in the latter country. In contrast, under autarky the utility in country 1 conditional on the pair of public finances at t = 1 being $\sigma = SC$ is given by:

$$U_1(\mathbf{1}|\sigma = SC) = pR - (1-p) \left[\int_0^{\lambda_S} k dG(k) + \lambda_S (1-G(\lambda_S)) \right].$$
(B16)

Using that $\hat{x}_{2,SC} > x^*$ and the fact that $k(x^*|2, SC) > \lambda_S$, which follows from Proposition 1, it follows that $\overline{k}(\hat{x}_{2,SC}|2, SC) > \lambda_S$. Using this inequality, (B15) and (B16), the inequality $U_1(\hat{\boldsymbol{x}}|\sigma = SC) > U_1(\mathbf{1}|\sigma = SC)$ can be written after some algebraic manipulation as:

$$\int_{0}^{\lambda_{S}} (1+k) \, dG(k) + \int_{\lambda_{S}}^{\overline{k}(\widehat{x}_{2,SC}|2,SC)} (1+\lambda_{S}) \, dG(k) + \int_{\overline{k}(\widehat{x}_{2,SC}|2,SC)}^{1} \widehat{x}_{2,SC}(1+\lambda_{S}) \, dG(k) > 0,$$

and is thus satisfied. This concludes the proof that when $\hat{x}_{1,SC} = 0$ we have that $U_1(\hat{x}) > U_1(\mathbf{1})$, contradicting the assumption that PC_1 is binding.

vi) If PC_2 is binding then \hat{x} satisfies: $\hat{x}_{2,SC} < x^* < \hat{x}_{1,CS} < 1, \hat{x}_{2,SS} < \hat{x}_{1,SS} = 1, \hat{x}_{2,CC} < \hat{x}_{1,CC} = 1, \hat{x}_{1,SC} = \hat{x}_{2,CS} = 1.$

The statements in this item extend those in iii)-v) and its proof is identical.

vii) PC_2 is not binding.

Suppose that PC_2 is binding. Taking vi) into consideration, considering the characterization in Proposition 1 of the ex-post optimal banking union \boldsymbol{x}^* , and given the inequalities $\frac{\partial U_1(\boldsymbol{x})}{\partial x_{1,\sigma}} < 0$ and $\frac{\partial U_1(\boldsymbol{x})}{\partial x_{2,\sigma}} > 0$, it follows that:

$$U_1(\widehat{\boldsymbol{x}}) < U_1(\boldsymbol{x}^*).$$

From this inequality, and using the fact that $q^{CS} < aq^{SC}$, Proposition 2 implies $U_1(\hat{x}) < \overline{U}_1$, which means that PC_1 is not binding.

To conclude the proof it suffices to notice that ii) and vi) imply that PC_1 is binding. Then, the statements in the proposition result from iii)-v).

viii) There exists a unique optimal banking union

Let \hat{x} and \hat{x}' denote two optimal banking unions. Then, they satisfy Properties *iii*)-v). Let μ and μ' be the Lagrange multiplier of the participation constraint of country 1 at \hat{x} and \hat{x}' , respectively. Using (B12), we have that:

$$\hat{x}_{2,SC} \le \hat{x}'_{2,SC}$$
 iff $\mu \le \mu'$ with equality iff $\mu = \mu'$. (B17)

We can conduct similar arguments in the other bank failure states in which there is positive foreign aid, namely in the states $(1, \sigma)$, with $\sigma \in \{SS, CC, SC\}$, and find that:

$$\widehat{x}'_{1,\sigma} \leq \widehat{x}_{1,\sigma} \text{ iff } \mu \leq \mu' \text{ with equality iff } \mu = \mu'.$$
 (B18)

We have that $U_1(\hat{x}') = U_1(\hat{x}) = \overline{U}_1$. Now, (B17) and (B18) imply that $U_1(\hat{x}') \ge U_1(\hat{x})$ iff $\mu \le \mu'$, with equality iff $\mu = \mu'$. It follows that $\mu = \mu'$ which, in turn, from (B17) and (B18) implies that $\hat{x} = \hat{x}'$.

Infeasibility of a banking union x with W(x) > W(1) when $q^{SC} = 1$

Suppose that $q^{SC} = 1$ and that there exists a banking union \boldsymbol{x} with $W(\boldsymbol{x}) > W(1)$. Let $\hat{\boldsymbol{x}}$ denote the optimal banking union which, a fortiori, satisfies $W(\hat{\boldsymbol{x}}) > W(1)$. Since country asymmetry is maximal, we have from Proposition 2 that $\hat{\boldsymbol{x}} \neq \boldsymbol{x}^*$, and at least one of the participation constraints is binding. If PC_2 is binding, then the properties of the derivatives of $W(\boldsymbol{x})$ and $U_2(\boldsymbol{x})$ with respect to $x_{1,SC}$, and the optimality of $\hat{\boldsymbol{x}}$, immediately imply that $\hat{x}_{1,SC} = 1$. Moreover, $W(\hat{\boldsymbol{x}}) > W(1)$ implies that $\hat{x}_{2,SC} < 1$. But then, it immediately follows that $U_1(\hat{\boldsymbol{x}}) < U_1(1)$.

We thus have that PC_1 is binding. The fact that $W(\hat{x}) > W(1)$ implies that $\hat{x}_{2,SC} \in (x^*, 1)$. From here the arguments in step v) of the proof of the case $q^{SC} < 1$ conducted above could be reproduced, which implies that $\hat{x}_{1,SC} = 1$. But this, again, would imply that $U_1(\hat{x}) < U_1(1)$.

Proof of Proposition 4 Using the characterization of the optimal banking union $\hat{x} \equiv (\hat{x}_{i,\sigma})$ in Proposition 3, we have that:

$$T(q^{CS} = 0) = (1 - p)q^{SC}(1 - \hat{x}_{2,SC}) \left(1 - G\left(\bar{k}(\hat{x}_{2,SC}|SC)\right)\right) - (25) - (1 - p)q^{SS}(1 - \hat{x}_{1,SS}) \left(1 - G\left(\bar{k}(\hat{x}_{1,SS}|SS)\right)\right).$$

Moreover, we can use the fact that the strong country's participation constraint is binding and write (12) as:

$$\underbrace{-(1+\lambda_S)T(q^{CS}=0)}_{Net \ cost \ of \ cross-country \ transfers \ to \ strong \ country} = \underbrace{(1-p)q^{SS}\left(\Pi(\widehat{x}_{1,SS}|1,SS)-\Pi^S\right)}_{Increase \ in \ agg. \ res. \ cost \ in \ state \ (1,SS)} > 0,$$
(26)

where we have used that $\hat{x}_{1,SS} < 1$, and the fact that $\Pi(x|1,SS)$ is strictly decreasing in x for x < 1. The latter inequality implies that the strong country is a net receiver of funds, that is $T(q^{CS} = 0) < 0$.

Proof of Proposition 5 The proof of this result is constructive. Consider a transfer m^* such that:

$$U_1(\boldsymbol{x}^*) + (1+\lambda_S)m^* \geq \overline{U}_1$$

$$U_2(\boldsymbol{x}^*) - (1+\lambda_S)m^* \geq \overline{U}_2.$$
(27)

Define the expected contribution of the strong country through this agreement as:

$$T(m^*) \equiv (1-p) \left(q^{SC} - q^{CS} \right) (1-x^*) \left(1 - G \left(\overline{k}(x^*) \right) \right) - m^*.$$

Then, we can write the inequalities in (27) as follows:

$$(1-p)q^{CS} (\Pi^{C} - \Pi(x^{*})) - (1+\lambda_{S})T(m^{*}) \geq 0$$

$$(1-p)q^{SC} (\Pi^{C} - \Pi(x^{*})) + (1+\lambda_{S})T(m^{*}) \geq 0.$$

We can therefore write the inequalities in (27) as:

$$-(1-p)q^{SC}\left(\Pi^{C}-\Pi(x^{*})\right) \leq (1+\lambda_{S})T(m^{*}) \leq (1-p)q^{CS}\left(\Pi^{C}-\Pi(x^{*})\right).$$

Recognizing that $q^{SC} > q^{CS}$, the latter inequality defines a non-empty interval $[m^*_{\min}, m^*_{\max}]$.

We now show that $m_{\min}^* > 0$ if and only if the ex-post optimal banking union is not feasible without initial transfers.

For the "if" part, assume that the ex-post optimal banking union is not feasible without initial transfers. Then, we know from the proof of Proposition 2 that $U_1(\boldsymbol{x}^*) < \overline{U}_1$. Hence, for the inequality $U_1(\boldsymbol{x}^*) + (1 + \lambda_S)m^* \geq \overline{U}_1$ to hold, we necessarily need $m^* > 0$.

For the "only if" part, assume for the sake of contradiction that both $U_1(\boldsymbol{x}^*) \geq \overline{U}_1$ and $U_2(\boldsymbol{x}^*) \geq \overline{U}_2$. If $U_1(\boldsymbol{x}^*) = \overline{U}_1$, then $U_1(\boldsymbol{x}^*) \geq \overline{U}_1$ is trivially satisfied for $m^* = 0$. If $U_1(\boldsymbol{x}^*) > \overline{U}_1$, then we also have that $U_2(\boldsymbol{x}^*) > \overline{U}_2$, since $q^{SC} > q^{CS}$. In this case, we can always find a sufficiently small $m^* < 0$ so that both $U_1(\boldsymbol{x}^*) \geq \overline{U}_1$ and $U_2(\boldsymbol{x}^*) \geq \overline{U}_2$ hold.

Proof of Proposition 6 The ex-post optimal banking union is feasible if and only if:

$$U_1(\boldsymbol{x}^*, \overline{\boldsymbol{p}}^*) - \overline{U}_1 \ge 0.$$
(28)

Denote the early intervention thresholds in autarky by $\overline{p}^{Aut} = \overline{p}(1|1, CS) = \overline{p}(1|2, SC)$. From (19), we have that $\overline{p}^{Aut} > \overline{p}(x^*|1, CS)$. Using the utility so that in (20), we can write the feasibility condition (28) as:

$$\begin{pmatrix} \int_{\overline{p}^{Aut}}^{1} (1-p_i) dF(p_i) \end{pmatrix} q^{CS} \left(\Pi^C - \Pi(x^*) \right) + \left(\int_{\overline{p}(x^*|1,CS)}^{\overline{p}^{Aut}} \left[pR - (1-p)\Pi(x^*) - L \right] dF(p) \right) q^{CS} \\ - \left(\int_{\overline{p}(x^*|1,CS)}^{1} (1-p_i) dF(p_i) \right) \left(q^{SC} - q^{CS} \right) (1-x^*) (1+\lambda_S) \left(1 - G\left(\overline{k}(x^*) \right) \right)$$

$$(29)$$

Notice the similarities with the analogous condition for the baseline model in (8). As in the proof of Proposition 2, it can be shown from the latter expression that there exist $a' \in (0, 1)$ such that $(\boldsymbol{x}^*, \boldsymbol{\overline{p}}^*)$ is feasible if and only if $q^{CS} \geq a' q^{SC}$.

Suppose for the rest of the proof that $q^{CS} < a'q^{SC}$ and $q^{SC} < 1$.

Let $(\boldsymbol{x}, \overline{\boldsymbol{p}})$ be a feasible banking union and let $W(\boldsymbol{x}, \overline{\boldsymbol{p}})$ and $Ui(\boldsymbol{x}, \overline{\boldsymbol{p}})$ denote the aggregate welfare function and the expected utility in country *i*, respectively. It is easy to check that the partial derivatives of these functions with respect to the $x_{i,\sigma}$ have the same properties as those of their baseline model analogous described at the beginning of the proof of Proposition 3. In addition, we have that, if $\frac{\partial U_i(\boldsymbol{x}, \overline{\boldsymbol{p}})}{\partial x_{i,\sigma}} \neq 0$, then the quotient $\frac{\frac{\partial W(\boldsymbol{x}, \overline{\boldsymbol{p}})}{\partial x_{i,\sigma}}}{\frac{\partial U_i(\boldsymbol{x}, \overline{\boldsymbol{p}})}{\partial x_{i,\sigma}}}$ is independent of $\overline{\boldsymbol{p}}$. Taking this into account, the statements i)-vi) in the proof of Proposition 3 can be proved with no changes in this extended set-up.

The following properties will be used frequently in the rest of the proof and can be easily

checked:

$$\frac{\partial W\left(\boldsymbol{x},\overline{\boldsymbol{p}}\right)}{\partial \overline{p}_{i,\sigma}}\Big|_{\overline{p}_{i,\sigma}=\overline{p}(x_{i,\sigma}|i,\sigma)} = 0, \text{ and } \frac{\partial W\left(\boldsymbol{x},\overline{\boldsymbol{p}}\right)}{\partial \overline{p}_{i,\sigma}} < 0 \text{ iff } \overline{p}_{i,\sigma} > \overline{p}(x_{i,\sigma}|i,\sigma) \quad (B19)$$

$$\frac{\partial U_{i}\left(\boldsymbol{x},\overline{\boldsymbol{p}}\right)}{\partial \overline{p}_{i,\sigma}} < 0 \text{ if } \overline{p}_{i,\sigma} > \overline{p}(x_{i,\sigma}|i,\sigma) \text{ or if } \overline{p}_{i,\sigma} = \overline{p}(x_{i,\sigma}|i,\sigma) \text{ and } x_{i,\sigma} < 1(30)$$

$$\frac{\partial U_{i}\left(\boldsymbol{x},\overline{\boldsymbol{p}}\right)}{\partial \overline{p}_{\overline{i},\sigma}} > 0 \text{ if } x_{\overline{i},\sigma} < 1, \text{ and } \frac{\partial U_{i}\left(\boldsymbol{x},\overline{\boldsymbol{p}}\right)}{\partial \overline{p}_{\overline{i},\sigma}} = 0 \text{ if } x_{\overline{i},\sigma} = 1.$$

Let us now prove that statement vii) in the proof of Proposition 3 remains valid, namely, PC_2 cannot be binding in the optimal banking union. Suppose, on the contrary, that PC_2 is binding. We argue that this leads to distortions relative to the policies in the ex-post optimal banking union $(\boldsymbol{x}^*, \boldsymbol{p}^*)$ that imply $U_1(\hat{\boldsymbol{x}}, \hat{\boldsymbol{p}}) < U_1(\boldsymbol{x}^*, \boldsymbol{p}^*)$ and, since by assumption the country asymmetry is large enough so that the banking union $(\boldsymbol{x}^*, \boldsymbol{p}^*)$ does not satisfy PC_1 , we would have a contradiction.

Since PC_2 is binding, we must have that $\hat{x}_{1,CS} > x^*$ and the ex-post optimal intervention threshold $\overline{p}(\hat{x}_{1,CS}|1,CS)$, conditional on $\hat{x}_{1,CS}$, satisfies $\overline{p}(\hat{x}_{1,CS}|1,CS) > p^*$. In addition, since PC_2 is binding, the properties in (B19) and the optimality of (\hat{x}, \hat{p}) imply that:

$$\widehat{\overline{p}}_{1,CS} > \overline{p}(\widehat{x}_{1,CS}|1,CS) > p^*.$$
(B20)

Similarly, for $\sigma = SS, CC, SC$, we have that $\hat{x}_{1,\sigma} = x_{1,\sigma}^* = 1$ and:

$$\overline{\overline{p}}_{1,\sigma} > \overline{p}(\widehat{x}_{1,\sigma}|1,\sigma) = \overline{p}(1|1,\sigma).$$
(B21)

It follows that the distortions in the resolution and early intervention policies of (\hat{x}, \hat{p}) , relative to (x^*, p^*) , in the bank failure states that involve the bank in country 1 strictly decrease the utility in that country.

Let us now consider the resolution and early intervention decisions of bank 2 in state $\sigma = SC$, that is $\hat{x}_{2,SC}$ and $\hat{\overline{p}}_{2,SC}$. In order to prove that the distortions induced by the fact that PC_2 is binding reduce the utility in country 1 it suffices to show that:

$$\left(\int_{\widehat{p}_{2,SC}}^{1} (1-p_2)dF(p_2)\right) q^{\sigma}t(\widehat{x}_{2,SC}|2,SC) \ge \left(\int_{\overline{p}^*}^{1} (1-p_2)dF(p_2)\right) q^{\sigma}t(x^*|2,SC), \quad (B22)$$

where the left hand side and the right hand side account for the social gross cost of the transfers from country 1 to country 2 under $(\hat{x}_{2,SC}, \overline{\hat{p}}_{2,SC})$ and (x^*, \overline{p}^*) , respectively. Suppose, on the contrary, that (B22) is not satisfied. Since we have that $\hat{x}_{2,SC} < x^*$, it has to be the case that $\hat{\overline{p}}_{2,SC} > \overline{p}^*$. Let $(\hat{x}', \hat{\overline{p}}')$ be the banking union that differs from $(\hat{x}, \hat{\overline{p}})$ only in the policies for bank 2 in the state $\sigma = SC$, which are defined to be $\hat{x}'_{2,SC} = x^*$ and $\hat{\overline{p}}'_{2,SC} = \overline{p}^*$.

The increase in $x_{2,SC}$ from $\hat{x}_{2,SC}$ to $\hat{x}'_{2,SC} = x^*$ is aggregate welfare increasing and, conditional on $\hat{x}'_{2,SC} = x^*$, the reduction in $\overline{p}_{2,SC}$ from $\hat{\overline{p}}_{2,SC}$ to $\hat{\overline{p}}'_{2,SC} = \overline{p}^*$ is also aggregate welfare increasing, which implies that $W(\hat{x}', \hat{\overline{p}}') > W(\hat{x}, \hat{\overline{p}})$. Moreover, the fact that (B22) is not satisfied means, by construction, that $U_2(\hat{x}', \hat{\overline{p}}') > U_2(\hat{x}, \hat{\overline{p}})$. The optimality of $(\hat{x}, \hat{\overline{p}})$ then implies that:

$$U_1(\widehat{\boldsymbol{x}}', \widehat{\overline{\boldsymbol{p}}}') < \overline{U}_1 < U_1(\widehat{\boldsymbol{x}}, \widehat{\overline{\boldsymbol{p}}}).$$

Notice that the increase in $x_{2,SC}$ from $\hat{x}_{2,SC}$ to $\hat{x}'_{2,SC} = x^*$ increases the utility in country 1, while the reduction in $\overline{p}_{2,SC}$ from $\hat{\overline{p}}_{2,SC}$ to $\hat{\overline{p}}'_{2,SC} = \overline{p}^*$ decreases it. As a result, the inequality above implies that there exists a banking union $(\hat{x}'', \hat{\overline{p}}'')$ that differs from $(\hat{x}, \hat{\overline{p}})$ only in the policies for bank 2 in state $\sigma = SC$, which are given by $\hat{x}'_{2,SC} = x^*$, $\hat{\overline{p}}'_{2,SC} \in (\overline{p}^*, \hat{\overline{p}}_{2,SC})$, and satisfy:

$$U_1(\widehat{\boldsymbol{x}}'', \widehat{\overline{\boldsymbol{p}}}'') = U_1(\widehat{\boldsymbol{x}}, \widehat{\overline{\boldsymbol{p}}}) \text{ and } W(\widehat{\boldsymbol{x}}'', \widehat{\overline{\boldsymbol{p}}}'') > W(\widehat{\boldsymbol{x}}, \widehat{\overline{\boldsymbol{p}}}).$$

The equality and inequality above in turn imply that $U_2(\widehat{\boldsymbol{x}}'', \widehat{\overline{\boldsymbol{p}}}'') > U_2(\widehat{\boldsymbol{x}}, \widehat{\overline{\boldsymbol{p}}}) = \overline{U}_2$. But then the banking union $(\widehat{\boldsymbol{x}}'', \widehat{\overline{\boldsymbol{p}}}'')$ would be feasible and thus optimal.

Analogous arguments show that the distortions in the policies for bank 2 in states $\sigma = SS, CC, CS$ induced by the fact that PC_2 is binding reduce the utility in country 1 relative to that under the ex-post optimal banking union. This concludes the proof that PC_2 cannot be binding.

We must thus have that PC_1 is binding. The statements in the proposition regarding the forbearance in early interventions of bank 1 and tightness in those of bank 2 are then an immediate consequence of the properties in (B19) and the optimality of $(\hat{x}, \hat{\overline{p}})$.

Proof of Proposition 7 An arbitrary funding mix $(x_{i,\sigma}, y_{i,\sigma})$ for bail-outs at country *i* in state σ consists of: the home contribution $x_{i,\sigma}$ at t = 1, the foreign loan $y_{i,\sigma}$ repaid by the home country at t = 2, and the foreign contribution $1 - x_{i,\sigma} - y_{i,\sigma}$ at t = 1. Notice that both home contributions and foreign loans provide incentives to the home authority not to overstate the need for bail-outs. In addition, by construction of the interest rate charged by the foreign country on loans, the expected utility of this country only depends on is direct foreign contribution.

Let $(\hat{\boldsymbol{x}}, \hat{\boldsymbol{y}})$ be an optimal banking union. In order to prove the proposition we will argue that for any bank failure state (i, σ) we have that either $\hat{x}_{i,\sigma} = 0$ or $\hat{y}_{i,\sigma} = 0$ and thus the optimal funding mix is characterized by a single variable $(\hat{y}_{i,\sigma} \text{ in states for which } \hat{x}_{i,\sigma} = 0,$ and $\hat{x}_{i,\sigma}$ in states for which $\hat{y}_{i,\sigma} = 0$). The results in the proposition then can be obtained by reproducing the arguments in the proofs of Propositions 1, 2 and 3.

i) If $(i, \sigma) \in \{(1, CS), (2, SC)\}$, then $\widehat{x}_{i,\sigma} = 0$.

Suppose that $\hat{x}_{i,\sigma} > 0$. Since $\lambda_C > \tilde{\lambda}_C$ and $R_S \leq 1$ (because $\tilde{\lambda}_S \geq \lambda_S$), we have that $1 + \lambda_C > (1 + \tilde{\lambda}_C) R_S$. This implies that the cost for the home country of home contributions is strictly higher than repaying foreign loans. Hence, the funding mix $(0, y'_{i,\sigma})$ such that $y'_{i,\sigma} = \hat{x}_{i,\sigma} + \hat{y}_{i,\sigma}$ strictly increases the home country's welfare relative to $(\hat{x}_{i,\sigma}, \hat{y}_{i,\sigma})$, without changing the foreign country's welfare. A banking union that only differs from (\hat{x}, \hat{y}) in the funding mix in the state (i, σ) , which is defined to be $(0, y'_{i,\sigma})$, would thus be feasible and increase aggregate welfare.

ii) If $(i, \sigma) \in \{(1, SC), (2, CS)\}$, then $\hat{y}_{i,\sigma} = 0$.

Suppose that $\hat{y}_{i,\sigma} > 0$. Since $\tilde{\lambda}_S \ge \lambda_S$ and $R_C > 1$ (because $\lambda_C > \tilde{\lambda}_C$), we have that $1 + \lambda_S < (1 + \tilde{\lambda}_S) R_C$. This implies that the cost for the home country of repaying foreign loans is strictly higher than that of home contributions. Hence, the funding mix $(x'_{i,\sigma}, 0)$ such that $x'_{i,\sigma} = \hat{x}_{i,\sigma} + \hat{y}_{i,\sigma}$ strictly increases the home country's welfare relative to $(\hat{x}_{i,\sigma}, \hat{y}_{i,\sigma})$, without changing the foreign country's welfare. A banking union that only differs from (\hat{x}, \hat{y}) in the funding mix in the state (i, σ) , which is defined to be $(x'_{i,\sigma}, 0)$, would thus be feasible and increase aggregate welfare.

iii) If $(i, \sigma) \in \{(1, SS), (1, CC), (2, SS), (2, CC)\}, \text{ then } \widehat{y}_{i,\sigma} = 0.$

Suppose that $\hat{y}_{i,\sigma} > 0$. We have that $1 + \lambda_S = (1 + \tilde{\lambda}_S) R_S$ and that $1 + \lambda_C = (1 + \tilde{\lambda}_C) R_C$. This implies that the cost for the home country of repaying foreign loans equals that of home contributions. Hence, the funding mix $(x'_{i,\sigma}, 0)$ such that $x'_{i,\sigma} = \hat{x}_{i,\sigma} + \hat{y}_{i,\sigma}$ provides the same welfare as $(\hat{x}_{i,\sigma}, \hat{y}_{i,\sigma})$ for the two countries. A banking union that only differs from (\hat{x}, \hat{y}) in the funding mix in the state (i, σ) , which is defined to be $(x'_{i,\sigma}, 0)$, is thus pay-off equivalent to (\hat{x}, \hat{y}) .

Proof of Proposition 8 We introduce some notation. For a given bank failure state (i, σ) and home contribution x we denote $\overline{k}^{\beta}(x|i, \sigma)$ the bail-out threshold in the economy with cross-border liabilities. We have that:

$$\overline{k}^{\beta}(x|i,\sigma) = \max\left\{ \left(1 + \lambda_{i,\sigma}\right) x - \left(1 - \beta\right), 0 \right\}$$

Consider (9), which is the characterization of the analogous threshold $\overline{k}(x|i,\sigma)$ in the baseline model. Then, for $x \geq \frac{\beta}{1+\lambda_{i,\sigma}}$ we have that:

$$\overline{k}^{\beta}(x - \frac{\beta}{1 + \lambda_{i,\sigma}}|i,\sigma) = \overline{k}^{0}(x|i,\sigma) = \overline{k}(x|i,\sigma).$$

Notice that for $x < \frac{\beta}{1+\lambda_{i,\sigma}}$ the equality also holds if we replace $x - \frac{\beta}{1+\lambda_{i,\sigma}}$ by max $\left\{x - \frac{\beta}{1+\lambda_{i,\sigma}}, 0\right\}$. For any country $j \in \{1, 2\}$, bank-failure tuple (i, σ) , and home contribution x, we denote

For any country $j \in \{1, 2\}$, bank-failure tuple (i, σ) , and home contribution x, we denote the overall utility at t = 1 of agents in country i when there are cross-border liabilities β as $U_i^{\beta}(x|i, \sigma)$. In addition, we denote $U_j(x|i, \sigma) = U_j^0(x|i, \sigma)$. Finally, we have that the overall utility as of t = 0 in country j associated with a banking union \boldsymbol{x} when there are cross-border liabilities β , which we denote by $U_j^{\beta}(\boldsymbol{x})$, satisfies:

$$U_{j}^{\beta}\left(\boldsymbol{x}\right) = \sum_{i,\sigma} q^{\sigma} U_{j}(x_{i,\sigma}|i,\sigma).$$
(31)

We can now proceed with the proof. Consider the optimal banking union \boldsymbol{x}^* of the baseline economy. We are going to construct a new banking union $\boldsymbol{x}' = (x'_{i,\sigma})$ satisfying for every j and (i, σ) the following inequality:

$$U_j^{\beta}(x_{i,\sigma}'|i,\sigma) - U_j^{\beta}(1|i,\sigma) \ge U_j(x_{i,\sigma}^*|i,\sigma) - U_j(1|i,\sigma),$$
(32)

with strict inequality for some i and (j, σ) . This will in particular imply that the banking union \mathbf{x}' is feasible in the economy with cross-border liabilities and that the welfare gains relative to autarky induced by \mathbf{x}' are strictly larger than those induced by \mathbf{x} in the baseline economy.

To streamline the expressions and exposition to follow, we make two simplifying assumptions. First, for all (i, σ) we have $\overline{k}^{\beta}(1|i, \sigma) > k_{\max}$ and thus there are no bail-outs in autarky in the cross-border economy. A necessary and sufficient condition for the assumption to be satisfied is that $\lambda_S + \beta > k_{\max}$. Second for all σ , we have $q^{\sigma} > 0$.

Consider the bank failure state (i, σ) . Suppose $\lambda_{i,\sigma} \geq \lambda_{\bar{i},\sigma}$. Let:

$$x_{i,\sigma}' = \max\left\{x_{i,\sigma}^* - \frac{\beta}{1 + \lambda_{i,\sigma}}, 0\right\}.$$
(33)

We have by construction that:

$$\overline{k}^{\beta}(x_{i,\sigma}'|i,\sigma) = \overline{k}(x_{i,\sigma}^{*}|i,\sigma), \qquad (34)$$

and that:

$$x_{i,\sigma}^* - x_{i,\sigma}' \le \frac{\beta}{1 + \lambda_{i,\sigma}}.$$
(35)

We have the following utility gains for the home country in the baseline and cross-border economies:

$$U_{i}(x_{i,\sigma}^{*}|i,\sigma) - U_{i}(1|i,\sigma) = \int_{\overline{k}(x_{i,\sigma}^{*}|i,\sigma)}^{\lambda_{i,\sigma}} \left(k - \overline{k}(x_{i,\sigma}^{*}|i,\sigma)\right) dG(k) + (1 - G(\lambda_{i,\sigma})) \left(\lambda_{i,\sigma} - \overline{k}(x_{i,\sigma}^{*}|i,\sigma)\right) dG(k) + (1 - G(\lambda_{i,\sigma})) \left(\lambda_{i,\sigma} - \overline{k}(x_{i,\sigma}^{*}|i,\sigma)\right) dG(k) dG(k) dG(k),$$

where we have used that $\overline{k}^{\beta}(1|i,\sigma) > k_{\max}$ and that $\overline{k}(1|i,\sigma) = \lambda_{i,\sigma}$. Using (34) it follows from the latter two expressions that:

$$U_{i}^{\beta}(x_{i,\sigma}'|i,\sigma) - U_{i}^{\beta}(1|i,\sigma) > U_{i}(x_{i,\sigma}^{*}|i,\sigma) - U_{i}(1|i,\sigma),$$

and (32) is strictly satisfied for j = i.

We have the following utility gains for the foreign country:

$$U_{\overline{i}}(x_{i,\sigma}^{*}|i,\sigma) - U_{\overline{i}}(1|i,\sigma) = -\left(1 - \overline{k}(x_{i,\sigma}|i,\sigma)\right)\left(1 + \lambda_{\overline{i},\sigma}\right)\left(1 - x_{i,\sigma}^{*}\right),$$

$$U_{\overline{i}}^{\beta}(x_{i,\sigma}'|i,\sigma) - U_{\overline{i}}^{\beta}(1|i,\sigma) = -\left(1 - \overline{k}^{\beta}(x_{i,\sigma}'|i,\sigma)\right)\left[\left(1 + \lambda_{\overline{i},\sigma}\right)\left(1 - x_{i,\sigma}'\right) - \beta\right],$$

where in the second expression we have used that when in any bail-out of bank *i* the measure β of debtholders in country \overline{i} are repaid and that in autarky there are no bail-outs. Taking into account (33), (34), (35), and the assumption that $\lambda_{i,\sigma} \geq \lambda_{\overline{i},\sigma}$, we have from the latter two expressions that:

$$\begin{split} U_{\overline{i}}^{\beta}(x_{i,\sigma}'|i,\sigma) - U_{\overline{i}}^{\beta}(1|i,\sigma) &= -\left(1 - \overline{k}(x_{i,\sigma}^{*}|i,\sigma)\right) \left[(1 + \lambda_{\overline{i},\sigma}) \left(1 - x_{i,\sigma}^{*}\right) - \beta \left(1 - (1 + \lambda_{\overline{i},\sigma}) \left(x_{i,\sigma}^{*} - x_{i,\sigma}'\right)\right) \right] \\ &\geq U_{\overline{i}}(x_{i,\sigma}^{*}|i,\sigma) - U_{\overline{i}}(1|i,\sigma) + \left(1 - \overline{k}(x_{i,\sigma}^{*}|i,\sigma)\right) \beta \frac{\lambda_{i,\sigma} - \lambda_{\overline{i},\sigma}}{1 + \lambda_{i,\sigma}} \geq \\ &\geq U_{\overline{i}}(x_{i,\sigma}^{*}|i,\sigma) - U_{\overline{i}}(1|i,\sigma), \end{split}$$

and (32) is satisfied for $j = \overline{i}$.

Suppose $\lambda_{i,\sigma} < \lambda_{\overline{i},\sigma}$, then Proposition 1 and 3 imply that the optimal banking union satisfies $x_{i,\sigma}^* = 1$. Let $x_{i,\sigma}' = 1$. Then we trivially have for any j = 1, 2 that:

$$U_j^{\beta}(x_{i,\sigma}'|i,\sigma) - U_j^{\beta}(1|i,\sigma) = U_j(x_{i,\sigma}^*|i,\sigma) - U_j(1|i,\sigma) = 0,$$

and (32) is satisfied for any j.

References

- Acharya, V. (2003). Is the international convergence of capital adequacy regulation desirable?. Journal of Finance, 58(6), 2745-2782.
- Agarwal, S., Lucca, D., Seru, A., & Trebbi, F. (2014). Inconsistent regulators: Evidence from banking. The Quarterly Journal of Economics, 129(2), 889-938.
- Beck, T., Da-Rocha-Lopes, S., & Silva, A. F., 2017. Sharing the pain? Credit supply and real effects of bank bail-ins. Centre for Banking Research Working Paper Series, 01/17.
- Beck, T., Wagner, W., 2016. Supranational supervision How much and for whom?. International Journal of Central Banking, 12(2), 221–268.
- Bian, B., Haselmann, R., Kick, T. and Vig, V., 2017. The political economy of bank bailouts. IMFS Working Paper Series.
- Brown, M., Evangelou, I., Stix, H., 2018. Banking Crises, Bail-ins and Money Holdings. Central Bank of Cyprus Working Paper No. 2017-2
- Bolton, P., Oehmke, M., 2018. Bank resolution and the structure of global banks. Review of Financial Studies, forthcoming.
- Calzolari, G., Colliard, J.-E., Loranth, G., 2018. Multinational banks and supranational supervision. Review of Financial Studies, forthcoming.
- Carletti, E., Dell'Ariccia, G., Marquez, R., 2016. Supervisory incentives in a banking union. IMF Working Paper No. 16/186.
- Claessens, S., van Horen, N., 2013. Impact of Foreign Banks. Journal of Financial Perspectives, 1(1), 29-42.
- Colliard, J.-E., 2018. Optimal supervisory architecture and financial integration in a banking union. HEC Paris Research Paper No. FIN-2018-1319.
- Colliard, J.-E., Gromb, D., 2018. Financial restructuring and resolution of banks. HEC Paris Research Paper No. FIN-2018-1272.
- Dell'Ariccia, G., Martinez Peria, M. S., Igan, D.O., Awadzi, E.A., Dobler, M., Sandri, D., 2018. Trade-offs in bank resolution. IMF Staff Discussion Note, February 2018.
- Dell'Ariccia, G., Marquez, R., 2006. Competition among regulators and credit market integration. Journal of Financial Economics 79(2), 401–430.

- Dewatripont, M., 2014. European banking: Bailout, bail-in and state aid control. International Journal of Industrial Organization, 34, 37-43.
- Farhi, E., & Werning, I. (2017). Fiscal unions. American Economic Review, 107(12), 3788-3834.
- Faia, E., Weder di Mauro, B., 2015. Cross-border resolution of global banks. SAFE Working Paper Series 88, Research Center SAFE - Sustainable Architecture for Finance in Europe, Goethe University Frankfurt.
- Ferrarini, G., 2015. Cross-border Banking and the SSM. European Economy, 2015.3 57.
- Foarta, O. D. (2018). The limits to partial banking unions: A political economy approach. American Economic Review 108(4-5): 1187–1213.
- Goodhart, C., & Avgouleas, E., 2016. Critical reflections on bank bail-ins. In The Reform of International Economic Governance (pp. 75-99). Routledge.
- Gopalan, Y., Kalda, A., Manela, A., 2017. Hub-and-spoke regulation and bank leverage. Working paper.
- Navaretti, G. B., Calzolari, G., Pozzolo, A. F., 2015. Sand in the wheels: Implementing the Single supervisory mechanism and multinational banking in Europe. European Economy, 1(3), 9-23.
- Schoenmaker, D., 2011. The financial trilemma. Economics letters, 111(1), 57-59.
- Stavrakeva, V., 2018. Optimal bank regulation and fiscal capacity. Review of Economic Studies, forthcoming.
- Tirole, J., 2015. Country solidarity in sovereign crises. American Economic Review, 105(8), 2333–63.
- Walther, A., White, L., 2017. Bail-ins and bail-outs in bank resolution. Working paper.

- N. 1188 Raising aspirations and higher education: evidence from the UK's Widening Participation Policy, by Lucia Rizzica (September 2018).
- N. 1189 *Nearly exact Bayesian estimation of non-linear no-arbitrage term structure models*, by Marcello Pericoli and Marco Taboga (September 2018).
- N. 1190 *Granular Sources of the Italian business cycle*, by Nicolò Gnocato and Concetta Rondinelli (September 2018).
- N. 1191 *Debt restructuring with multiple bank relationships*, by Angelo Baglioni, Luca Colombo and Paola Rossi (September 2018).
- N.1192 Exchange rate pass-through into euro area inflation. An estimated structural model, by Lorenzo Burlon, Alessandro Notarpietro and Massimiliano Pisani (September 2018).
- N. 1193 *The effect of grants on university drop-out rates: evidence on the Italian case*, by Francesca Modena, Enrico Rettore and Giulia Martina Tanzi (September 2018).
- N. 1194 *Potential output and microeconomic heterogeneity*, by Davide Fantino (November 2018).
- N. 1195 Immigrants, labor market dynamics and adjustment to shocks in the Euro Area, by Gaetano Basso, Francesco D'Amuri and Giovanni Peri (November 2018).
- N. 1196 Sovereign debt maturity structure and its costs, by Flavia Corneli (November 2018).
- N. 1197 Fiscal policy in the US: a new measure of uncertainty and its recent development, by Alessio Anzuini and Luca Rossi (November 2018).
- N.1198 Macroeconomics determinants of the correlation between stocks and bonds, by Marcello Pericoli (November 2018).
- N. 1199 Bank capital constraints, lending supply and economic activity, by Antonio M. Conti, Andrea Nobili and Federico M. Signoretti (November 2018).
- N. 1200 *The effectiveness of capital controls*, by Valerio Nispi Landi and Alessandro Schiavone (November 2018).
- N. 1201 Contagion in the CoCos market? A case study of two stress events, by Pierluigi Bologna, Arianna Miglietta and Anatoli Segura (November 2018).
- N. 1202 Is ECB monetary policy more powerful during expansions?, by Martina Cecioni (December 2018).
- N. 1203 *Firms' inflation expectations and investment plans*, by Adriana Grasso and Tiziano Ropele (December 2018).
- N. 1204 *Recent trends in economic activity and TFP in Italy with a focus on embodied technical progress*, by Alessandro Mistretta and Francesco Zollino (December 2018).
- N. 1205 Benefits of Gradualism or Costs of Inaction? Monetary Policy in Times of Uncertainty, by Giuseppe Ferrero, Mario Pietrunti and Andrea Tiseno (February 2019).
- N. 1206 Machine learning in the service of policy targeting: the case of public credit guarantees, by Monica Andini, Michela Boldrini, Emanuele Ciani, Guido de Blasio, Alessio D'Ignazio and Andrea Paladini (February 2019).
- N. 1207 Do the ECB's monetary policies benefit Emerging Market Economies? A GVAR analysis on the crisis and post-crisis period, by Andrea Colabella (February 2019).
- N. 1208 The Economic Effects of Big Events: Evidence from the Great Jubilee 2000 in Rome, by Raffaello Bronzini, Sauro Mocetti and Matteo Mongardini (February 2019).
- N. 1209 *The added value of more accurate predictions for school rankings*, by Fritz Schiltz, Paolo Sestito, Tommaso Agasisti and Kristof De Witte (February 2019).

^(*) Requests for copies should be sent to:

Banca d'Italia – Servizio Studi di struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

2017

- AABERGE, R., F. BOURGUIGNON, A. BRANDOLINI, F. FERREIRA, J. GORNICK, J. HILLS, M. JÄNTTI, S. JENKINS, J. MICKLEWRIGHT, E. MARLIER, B. NOLAN, T. PIKETTY, W. RADERMACHER, T. SMEEDING, N. STERN, J. STIGLITZ, H. SUTHERLAND, *Tony Atkinson and his legacy*, Review of Income and Wealth, v. 63, 3, pp. 411-444, WP 1138 (September 2017).
- ACCETTURO A., M. BUGAMELLI and A. LAMORGESE, *Law enforcement and political participation: Italy 1861-*65, Journal of Economic Behavior & Organization, v. 140, pp. 224-245, **WP 1124 (July 2017).**
- ADAMOPOULOU A. and G.M. TANZI, *Academic dropout and the great recession*, Journal of Human Capital, V. 11, 1, pp. 35–71, **WP 970** (October 2014).
- ALBERTAZZI U., M. BOTTERO and G. SENE, *Information externalities in the credit market and the spell of credit rationing*, Journal of Financial Intermediation, v. 30, pp. 61–70, WP 980 (November 2014).
- ALESSANDRI P. and H. MUMTAZ, *Financial indicators and density forecasts for US output and inflation*, Review of Economic Dynamics, v. 24, pp. 66-78, **WP 977** (November 2014).
- BARBIERI G., C. ROSSETTI and P. SESTITO, *Teacher motivation and student learning*, Politica economica/Journal of Economic Policy, v. 33, 1, pp.59-72, WP 761 (June 2010).
- BENTIVOGLI C. and M. LITTERIO, Foreign ownership and performance: evidence from a panel of Italian firms, International Journal of the Economics of Business, v. 24, 3, pp. 251-273, WP 1085 (October 2016).
- BRONZINI R. and A. D'IGNAZIO, *Bank internationalisation and firm exports: evidence from matched firmbank data*, Review of International Economics, v. 25, 3, pp. 476-499 WP 1055 (March 2016).
- BRUCHE M. and A. SEGURA, *Debt maturity and the liquidity of secondary debt markets*, Journal of Financial Economics, v. 124, 3, pp. 599-613, WP 1049 (January 2016).
- BURLON L., *Public expenditure distribution, voting, and growth,* Journal of Public Economic Theory,, v. 19, 4, pp. 789–810, **WP 961 (April 2014).**
- BURLON L., A. GERALI, A. NOTARPIETRO and M. PISANI, Macroeconomic effectiveness of non-standard monetary policy and early exit. a model-based evaluation, International Finance, v. 20, 2, pp.155-173, WP 1074 (July 2016).
- BUSETTI F., *Quantile aggregation of density forecasts*, Oxford Bulletin of Economics and Statistics, v. 79, 4, pp. 495-512, **WP 979 (November 2014).**
- CESARONI T. and S. IEZZI, *The predictive content of business survey indicators: evidence from SIGE,* Journal of Business Cycle Research, v.13, 1, pp 75–104, WP 1031 (October 2015).
- CONTI P., D. MARELLA and A. NERI, Statistical matching and uncertainty analysis in combining household income and expenditure data, Statistical Methods & Applications, v. 26, 3, pp 485–505, WP 1018 (July 2015).
- D'AMURI F., *Monitoring and disincentives in containing paid sick leave*, Labour Economics, v. 49, pp. 74-83, **WP 787 (January 2011).**
- D'AMURI F. and J. MARCUCCI, *The predictive power of google searches in forecasting unemployment*, International Journal of Forecasting, v. 33, 4, pp. 801-816, **WP 891 (November 2012).**
- DE BLASIO G. and S. POY, *The impact of local minimum wages on employment: evidence from Italy in the* 1950s, Journal of Regional Science, v. 57, 1, pp. 48-74, **WP 953 (March 2014).**
- DEL GIOVANE P., A. NOBILI and F. M. SIGNORETTI, Assessing the sources of credit supply tightening: was the sovereign debt crisis different from Lehman?, International Journal of Central Banking, v. 13, 2, pp. 197-234, WP 942 (November 2013).
- DEL PRETE S., M. PAGNINI, P. ROSSI and V. VACCA, Lending organization and credit supply during the 2008–2009 crisis, Economic Notes, v. 46, 2, pp. 207–236, WP 1108 (April 2017).
- DELLE MONACHE D. and I. PETRELLA, Adaptive models and heavy tails with an application to inflation forecasting, International Journal of Forecasting, v. 33, 2, pp. 482-501, WP 1052 (March 2016).
- FEDERICO S. and E. TOSTI, *Exporters and importers of services: firm-level evidence on Italy*, The World Economy, v. 40, 10, pp. 2078-2096, **WP 877 (September 2012).**
- GIACOMELLI S. and C. MENON, *Does weak contract enforcement affect firm size? Evidence from the neighbour's court,* Journal of Economic Geography, v. 17, 6, pp. 1251-1282, WP 898 (January 2013).
- LOBERTO M. and C. PERRICONE, *Does trend inflation make a difference?*, Economic Modelling, v. 61, pp. 351–375, **WP 1033 (October 2015).**

- MANCINI A.L., C. MONFARDINI and S. PASQUA, Is a good example the best sermon? Children's imitation of parental reading, Review of Economics of the Household, v. 15, 3, pp 965–993, D No. 958 (April 2014).
- MEEKS R., B. NELSON and P. ALESSANDRI, *Shadow banks and macroeconomic instability*, Journal of Money, Credit and Banking, v. 49, 7, pp. 1483–1516, **WP 939** (November 2013).
- MICUCCI G. and P. ROSSI, *Debt restructuring and the role of banks' organizational structure and lending technologies*, Journal of Financial Services Research, v. 51, 3, pp 339–361, **WP 763 (June 2010)**.
- MOCETTI S., M. PAGNINI and E. SETTE, *Information technology and banking organization*, Journal of Journal of Financial Services Research, v. 51, pp. 313-338, WP 752 (March 2010).
- MOCETTI S. and E. VIVIANO, *Looking behind mortgage delinquencies*, Journal of Banking & Finance, v. 75, pp. 53-63, **WP 999 (January 2015).**
- NOBILI A. and F. ZOLLINO, *A structural model for the housing and credit market in Italy*, Journal of Housing Economics, v. 36, pp. 73-87, **WP 887 (October 2012).**
- PALAZZO F., Search costs and the severity of adverse selection, Research in Economics, v. 71, 1, pp. 171-197, WP 1073 (July 2016).
- PATACCHINI E. and E. RAINONE, Social ties and the demand for financial services, Journal of Financial Services Research, v. 52, 1–2, pp 35–88, WP 1115 (June 2017).
- PATACCHINI E., E. RAINONE and Y. ZENOU, *Heterogeneous peer effects in education*, Journal of Economic Behavior & Organization, v. 134, pp. 190–227, WP 1048 (January 2016).
- SBRANA G., A. SILVESTRINI and F. VENDITTI, *Short-term inflation forecasting: the M.E.T.A. approach*, International Journal of Forecasting, v. 33, 4, pp. 1065-1081, **WP 1016 (June 2015).**
- SEGURA A. and J. SUAREZ, How excessive is banks' maturity transformation?, Review of Financial Studies, v. 30, 10, pp. 3538–3580, WP 1065 (April 2016).
- VACCA V., An unexpected crisis? Looking at pricing effectiveness of heterogeneous banks, Economic Notes, v. 46, 2, pp. 171–206, WP 814 (July 2011).
- VERGARA CAFFARELI F., One-way flow networks with decreasing returns to linking, Dynamic Games and Applications, v. 7, 2, pp. 323-345, **WP 734** (November 2009).
- ZAGHINI A., A Tale of fragmentation: corporate funding in the euro-area bond market, International Review of Financial Analysis, v. 49, pp. 59-68, WP 1104 (February 2017).

2018

- ADAMOPOULOU A. and E. KAYA, *Young adults living with their parents and the influence of peers*, Oxford Bulletin of Economics and Statistics, v. 80, pp. 689-713, **WP 1038** (November 2015).
- ANDINI M., E. CIANI, G. DE BLASIO, A. D'IGNAZIO and V. SILVESTRINI, *Targeting with machine learning:* an application to a tax rebate program in Italy, Journal of Economic Behavior & Organization, v. 156, pp. 86-102, WP 1158 (December 2017).
- BARONE G., G. DE BLASIO and S. MOCETTI, *The real effects of credit crunch in the great recession: evidence from Italian provinces*, Regional Science and Urban Economics, v. 70, pp. 352-59, **WP 1057** (March 2016).
- BELOTTI F. and G. ILARDI Consistent inference in fixed-effects stochastic frontier models, Journal of Econometrics, v. 202, 2, pp. 161-177, WP 1147 (October 2017).
- BERTON F., S. MOCETTI, A. PRESBITERO and M. RICHIARDI, *Banks, firms, and jobs,* Review of Financial Studies, v.31, 6, pp. 2113-2156, **WP 1097 (February 2017).**
- BOFONDI M., L. CARPINELLI and E. SETTE, *Credit supply during a sovereign debt crisis*, Journal of the European Economic Association, v.16, 3, pp. 696-729, **WP 909** (April 2013).
- BOKAN N., A. GERALI, S. GOMES, P. JACQUINOT and M. PISANI, EAGLE-FLI: a macroeconomic model of banking and financial interdependence in the euro area, Economic Modelling, v. 69, C, pp. 249-280, WP 1064 (April 2016).
- BRILLI Y. and M. TONELLO, Does increasing compulsory education reduce or displace adolescent crime? New evidence from administrative and victimization data, CESifo Economic Studies, v. 64, 1, pp. 15–4, WP 1008 (April 2015).

- BUONO I. and S. FORMAI *The heterogeneous response of domestic sales and exports to bank credit shocks,* Journal of International Economics, v. 113, pp. 55-73, WP 1066 (March 2018).
- BURLON L., A. GERALI, A. NOTARPIETRO and M. PISANI, Non-standard monetary policy, asset prices and macroprudential policy in a monetary union, Journal of International Money and Finance, v. 88, pp. 25-53, WP 1089 (October 2016).
- CARTA F. and M. DE PHLIPPIS, You've Come a long way, baby. Husbands' commuting time and family labour supply, Regional Science and Urban Economics, v. 69, pp. 25-37, WP 1003 (March 2015).
- CARTA F. and L. RIZZICA, *Early kindergarten, maternal labor supply and children's outcomes: evidence from Italy*, Journal of Public Economics, v. 158, pp. 79-102, WP 1030 (October 2015).
- CASIRAGHI M., E. GAIOTTI, L. RODANO and A. SECCHI, A "Reverse Robin Hood"? The distributional implications of non-standard monetary policy for Italian households, Journal of International Money and Finance, v. 85, pp. 215-235, WP 1077 (July 2016).
- CECCHETTI S., F. NATOLI and L. SIGALOTTI, *Tail co-movement in inflation expectations as an indicator of anchoring*, International Journal of Central Banking, v. 14, 1, pp. 35-71, **WP 1025** (July 2015).
- CIANI E. and C. DEIANA, *No Free lunch, buddy: housing transfers and informal care later in life*, Review of Economics of the Household, v.16, 4, pp. 971-1001, **WP 1117 (June 2017).**
- CIPRIANI M., A. GUARINO, G. GUAZZAROTTI, F. TAGLIATI and S. FISHER, *Informational contagion in the laboratory*, Review of Finance, v. 22, 3, pp. 877-904, **WP 1063 (April 2016).**
- DE BLASIO G, S. DE MITRI, S. D'IGNAZIO, P. FINALDI RUSSO and L. STOPPANI, *Public guarantees to SME borrowing. A RDD evaluation*, Journal of Banking & Finance, v. 96, pp. 73-86, **WP 1111 (April 2017).**
- GERALI A., A. LOCARNO, A. NOTARPIETRO and M. PISANI, *The sovereign crisis and Italy's potential output*, Journal of Policy Modeling, v. 40, 2, pp. 418-433, **WP 1010** (June 2015).
- LIBERATI D., An estimated DSGE model with search and matching frictions in the credit market, International Journal of Monetary Economics and Finance (IJMEF), v. 11, 6, pp. 567-617, WP 986 (November 2014).
- LINARELLO A., Direct and indirect effects of trade liberalization: evidence from Chile, Journal of Development Economics, v. 134, pp. 160-175, WP 994 (December 2014).
- NUCCI F. and M. RIGGI, *Labor force participation, wage rigidities, and inflation,* Journal of Macroeconomics, v. 55, 3 pp. 274-292, WP 1054 (March 2016).
- RIGON M. and F. ZANETTI, *Optimal monetary policy and fiscal policy interaction in a non_ricardian economy,* International Journal of Central Banking, v. 14 3, pp. 389-436, WP 1155 (December 2017).
- SEGURA A., Why did sponsor banks rescue their SIVs?, Review of Finance, v. 22, 2, pp. 661-697, WP 1100 (February 2017).

2019

CIANI E. and P. FISHER, *Dif-in-dif estimators of multiplicative treatment effects*, Journal of Econometric Methods, v. 8. 1, pp. 1-10, **WP 985** (November 2014).

FORTHCOMING

- ACCETTURO A., W. DI GIACINTO, G. MICUCCI and M. PAGNINI, *Geography, productivity and trade: does* selection explain why some locations are more productive than others?, Journal of Regional Science, **WP 910 (April 2013).**
- ALBANESE G., G. DE BLASIO and P. SESTITO, *Trust, risk and time preferences: evidence from survey data,* International Review of Economics, **WP 911 (April 2013).**
- APRIGLIANO V., G. ARDIZZI and L. MONTEFORTE, Using the payment system data to forecast the economic activity, International Journal of Central Banking, WP 1098 (February 2017).
- ARNAUDO D., G. MICUCCI, M. RIGON and P. ROSSI, *Should I stay or should I go? Firms' mobility across banks in the aftermath of the financial crisis*, Italian Economic Journal / Rivista italiana degli economisti, **WP 1086 (October 2016).**

- BELOTTI F. and G. ILARDI, *Consistent inference in fixed-effects stochastic frontier models*, Journal of Econometrics, WP 1147 (October 2017).
- BUSETTI F. and M. CAIVANO, Low frequency drivers of the real interest rate: empirical evidence for advanced economies, International Finance, WP 1132 (September 2017).
- CHIADES P., L. GRECO, V. MENGOTTO, L. MORETTI and P. VALBONESI, Fiscal consolidation by intergovernmental transfers cuts? Economic Modelling, WP 1076 (July 2016).
- CIANI E., F. DAVID and G. DE BLASIO, *Local responses to labor demand shocks: a re-assessment of the case of Italy*, IMF Economic Review, **WP 1112 (April 2017).**
- COLETTA M., R. DE BONIS and S. PIERMATTEI, Household debt in OECD countries: the role of supply-side and demand-side factors, Social Indicators Research, WP 989 (November 2014).
- CORSELLO F. and V. NISPI LANDI, *Labor market and financial shocks: a time-varying analysis*, Journal of Money, Credit and Banking, **WP 1179 (June 2018).**
- COVA P., P. PAGANO and M. PISANI, Domestic and international macroeconomic effects of the Eurosystem Expanded Asset Purchase Programme, IMF Economic Review, WP 1036 (October 2015).
- D'AMURI F., Monitoring and disincentives in containing paid sick leave, Labour Economics, WP 787 (January 2011).
- D'IGNAZIO A. and C. MENON, *The causal effect of credit Guarantees for SMEs: evidence from Italy*, Scandinavian Journal of Economics, **WP 900 (February 2013).**
- ERCOLANI V. and J. VALLE E AZEVEDO, *How can the government spending multiplier be small at the zero lower bound?*, Macroeconomic Dynamics, **WP 1174 (April 2018).**
- FEDERICO S. and E. TOSTI, *Exporters and importers of services: firm-level evidence on Italy*, The World Economy, **WP 877 (September 2012).**
- GERALI A. and S. NERI, *Natural rates across the Atlantic*, Journal of Macroeconomics, WP 1140 (September 2017).
- GIACOMELLI S. and C. MENON, *Does weak contract enforcement affect firm size? Evidence from the neighbour's court,* Journal of Economic Geography, **WP 898 (January 2013).**
- GIORDANO C., M. MARINUCCI and A. SILVESTRINI, *The macro determinants of firms' and households' investment: evidence from Italy*, Economic Modelling, WP 1167 (March 2018).
- NATOLI F. and L. SIGALOTTI, *Tail co-movement in inflation expectations as an indicator of anchoring,* International Journal of Central Banking, WP 1025 (July 2015).
- RIGGI M., Capital destruction, jobless recoveries, and the discipline device role of unemployment, Macroeconomic Dynamics, WP 871 (July 2012).
- RIZZICA L., Raising aspirations and higher education. evidence from the UK's widening participation policy, Journal of Labor Economics, WP 1188 (September 2018).
- SEGURA A., Why did sponsor banks rescue their SIVs?, Review of Finance, WP 1100 (February 2017).